



Review Flexible Mobility On-Demand: An Environmental Scan

Sohani Liyanage *¹, Hussein Dia *¹, Rusul Abduljabbar and Saeed Asadi Bagloee

Department of Civil and Construction Engineering, Swinburne University of Technology, PO Box 218 Hawthorn, Australia; rabduljabbar@swin.edu.au (R.A.); sasadibagloee@swin.edu.au (S.A.B.)

* Correspondence: sliyanage@swin.edu.au (S.L.); hdia@swin.edu.au (H.D.);

Tel.: +61-392-144-336 (S.L.); +61-392-145-280 (H.D.)

Received: 22 January 2019; Accepted: 22 February 2019; Published: 27 February 2019



Abstract: On-demand shared mobility is increasingly being promoted as an influential strategy to address urban transport challenges in large and fast-growing cities. The appeal of this form of transport is largely attributed to its convenience, ease of use, and affordability made possible through digital platforms and innovations. The convergence of the shared economy with a number of established and emerging technologies—such as artificial intelligence (AI), Internet of Things (IoT), and Cloud and Fog computing—is helping to expedite their deployment as a new form of public transport. Recently, this has manifested itself in the form of Flexible Mobility on Demand (FMoD) solutions, aimed at meeting personal travel demands through flexible routing and scheduling. Increasingly, these shared mobility solutions are blurring the boundaries with existing forms of public transport, particularly bus operations. This paper presents an environmental scan and analysis of the technological, social, and economic impacts surrounding disruptive technology-driven shared mobility trends. Specifically, the paper includes an examination of current and anticipated external factors that are of direct relevance to collaborative and low carbon mobility. The paper also outlines how these trends are likely to influence the mobility industries now and into the future. The paper collates information from a wide body of literature and reports on findings from actual 'use cases' that exist today which have used these disruptive mobility solutions to deliver substantial benefits to travellers around the world. Finally, the paper provides stakeholders with insight into identifying and responding to the likely needs and impacts of FMoD and informs their policy and strategy positions on the implementation of smart mobility systems in their cities and jurisdictions.

Keywords: Flexible Mobility on Demand (FMoD); Mobility-as-a-Service (MaaS); shared mobility; Internet of Things (IoT); Cloud and Fog computing; sustainable public transport

1. Introduction

Providing access to high-quality urban transport services requires a variety of planning and operational innovations, as well as a better understanding of travel behaviour, operational processes, and the factors, which affect these issues.

Cities around the world are increasingly becoming a complex network of systems (also called super networks [1]) that are instrumented and interconnected in providing an opportunity for better management of vital services such as transport. An "Internet of Things (IoT)" is comprised of sensors and mobile devices all communicating with each other to enhance infrastructure capability and resilience and capturing volumes of data. Through data mining, artificial intelligence, and predictive analytics, smart city systems can provide travellers with more options to meet their mobility needs and allow city managers to monitor the performance of vital infrastructure and identify key areas where city services are lagging.

The convergence of physical and digital worlds is creating unprecedented opportunities to enhance the travel experience for millions of people every day through new mobility solutions driven by disruptive forces. Although some of these disruptive forces are still a few years away (e.g., autonomous vehicles), they have already started to shape a vision for a mobility transformation driven by converging forces including vehicle electrification, automated self-driving, mobile computing, on-demand shared mobility services, Big Data and predictive analytics [2]. The coming together of these powerful trends are shaping an urban mobility future inspired by a vision of low carbon living and zero road injuries. However, key to the success of these systems is a good understanding of the new business models, enabling technologies, and policies and regulations.

1.1. The Changing Landscape of Urban Mobility: Access Versus Ownership and the Collaborative Economy

The disruptive mobility trends have the potential to change fundamentally the relationship between the consumer and automobile. The rise of the collaborative or sharing economy, popularised by the companies such as Airbnb, Zipcar and Uber, has enjoyed remarkable rapid growth over the last few years and looks set to scale new heights over the next decade [3]. Access to mobility rather than to car ownership will enable customers to be more selective in choosing from the door-to-door mobility services offered by 'mobility operators' for intercity, suburban as well as 'last kilometre' travel solutions. Today, consumers use their cars as 'all-purpose' vehicles regardless of whether they are commuting alone to work or enjoying a leisurely drive with the family to the beach. There are already significant early signs that the importance of private car ownership is declining, and shared/collaborative mobility is increasing. With mobility offered as a service, consumers in the future will have the flexibility to choose the best solution for a specific trip purpose, on demand using their smartphones. According to a recent report [3], the shift to diverse mobility solutions is likely to result in 10% of new cars sold in 2030 to be a shared vehicle. This will reduce private-use vehicle sales but at the same time increase utilisation of shared vehicles. This will lead to a faster replacement rate of shared vehicles, which may offset some of the reduction in private-use vehicles. The report also estimates that more than 30% of kilometres driven in new cars sold could be from shared mobility.

1.2. The Need for Shared On-Demand Mobility Solutions

Today, with ever-rising urbanisation, most cities around the world are challenged to provide transport infrastructure and mobility services to meet the travel demands of increasingly connected populations [4–9]. The role of reliable public transport services becomes crucial as this leads to sustainable urban mobility where the movement of people, goods, and freight is safer and more efficient. However, conventional public transport services operate in fixed routes and schedules, which make them less attractive compared to personalised travel options such as private modes, which offer more convenience and flexibility. The low frequency during off-peak periods and in low-density areas also reduces the reliability of conventional public transport [10]. This leads to car-dependent communities, where there are higher numbers of low occupancy vehicles on the road. This leads to higher congestion, energy consumption, and increases in emissions and environmental pollution [11–13].

Flexible on-demand transport has increasingly become attractive as a sustainable and economically feasible alternative to conventional bus services [14–16]. Flexible Mobility on Demand (FMoD) systems, also known as demand-responsive transport (DRT), Dial-a-Ride Transit (DART), and flexible transport services (FTS), have been trialled with variable success across a number of cities around the world. In this paper, FMoD refers to both shared mobility solutions of passenger vehicles and also the emerging forms of on-demand public transport that use larger vehicles such as small buses. In the last decade or so, door-to-door transport services have evolved from DRT services, which operate as a service for elderly and disabled users who have difficulties in using conventional public transport [17–19], to the more sophisticated app-based services in operation today. Interestingly, these forms of shared transport are gaining more popularity. For example, a survey in the San Francisco Bay

Area showed that 60% of respondents were willing to accept these personalised services [20]. From a transport service operator perspective, fixed-route and fixed-schedule conventional bus services may appear to be cost-effective in urban areas, but they have been found to perform poorly in low-density areas [21] and outside peak hours. Hence, there has been some strong and renewed interest to explore how demand-responsive transport would be beneficial for both users and operators. This is reflected in recent research in this field, which has not only looked at trials but also at the potential benefits of FMoD as a form of future urban mobility.

FMoD can be thought of as a form of public transport which aims to provide a convenient service to users through advanced technologies [22]. It is a user-targeted innovative approach aimed at providing efficient, safe, and economical mobility options to passengers and users of the transport system [23]. Others think of it as the transformation of private automobile ownership models towards more flexible shared solutions [23]. To be successful, FMoD should integrate and connect transit networks and operations, in real-time data, and provide users with a variety of new travel options. This allows users to plan and book their trips, get access to real-time information and to process the payment through a single user interface [24].

In its early days, users had to call and place their request few days prior to the trip, and then operators manually scheduled the trip [17]. However, advancements in mobile technologies and communications have led to rapid advances in Flexible Mobility on Demand services [22]. Users are now able to request, place their booking online and track the vehicles in real-time via the smartphone application. The size of vehicles has also changed, making way for more compact vehicles suited for transporting smaller numbers of people. In the process, this had led to a blurring of the lines between traditional public transport (buses) and shared mobility services. The characteristics of this user-oriented public transport include flexible routing and scheduling of adequate fleets that need to be operated in the form of shared transport as per passengers' requests on pick-up and drop-off locations. Therefore, the main ethos of a successful FMoD scheme is shared services, reliability, affordability, accessibility, and on-demand In addition, these FMoD are flexible in terms of route choice, vehicle allocation, and payment type [25]. Specifically, the main principle behind the FMoD operation is that users share their rides with passengers who follow similar routes from origin to destination. In terms of accessibility, customers with special needs such as disabilities, mobility aids, or other additional assistance could request rides suiting their needs.

1.3. Technological, Social, and Economic Impacts of Shared On-Demand Mobility Solutions

This paper focuses on providing an overview of flexible on-demand transport with emphasis on the role of these services as a form of public transport. The paper also looks more broadly at the technological, social, and economic impacts of shared on-demand mobility solutions through a comprehensive environmental scan which included acquisition and use of information about the topic in addition to developing a good understanding of the trends and relationships related to the topic. The paper's finding will address the needs of different stakeholders involved in shared mobility including researchers, practitioners, and policy makers [26]. For the environmental scan in this paper, the main search terms used to gather the information included; Mobility-as-a-Service (MaaS), demand-responsive transportation (DRT), flexible transport services (FTS), Dial-a-Ride Transit (DART), Mobility on Demand (MoD), Autonomous Mobility on Demand (AMoD), Flexible Mobility on Demand (FMoD), customized bus service, full demand service, etc. The main research databases that were used to gather and collate the information included Google Scholar, Research Gate, Springer, Scopus, Science Direct, TRB, IEEE, etc. The paper also benefited from findings from national and international stakeholder workshops attended by a number of authors.

The remainder of this paper is organised as follows: Section 2 provides a detailed overview of Flexible Mobility on Demand (FMoD) systems. Section 2 is subdivided as follows: (i) the historical perspective: early beginnings, and (ii) the case studies—mapping the value beyond the hype. Section 3 summarises the impacts, benefits, and user acceptance of shared mobility. Section 4 underlying

the disruptive technologies in FMoD. The disruptive technologies in FMoD that are discussed in Section 4 are digitalisation and the internet of things, mobile computing, Internet of Things (IoT), cloud, fog and quantum computing, crowd-sourcing and data fusion, and data analysis techniques. The summarised data analysis techniques are linear regression, time series analysis, clustering and classification methods, machine learning, big data analytics, optimisation techniques, deep learning and reinforcement learning. Then Section 5 discusses the existing and emerging business models. In Section 6 the challenges and opportunities of FMoD are described. Section 7 describes policy insights and Section 8 provides a summary of this paper review and future research directions.

2. Flexible Mobility on Demand (FMoD)

This novel user-centric approach focuses on providing convenient mobility options. The goal is to enable a more efficient, safer, reliable, and smarter mobility to benefit individual travellers, transport system operators and managers within a multimodal ecosystem. The key advantage of FMoD is in providing users with enhanced travel options through integrated, efficient, user-focused transport system. Private and public transport providers also benefit through integrated and common mobility service platforms.

2.1. Historical Perspective: Early Beginnings

While recent advances and breakthroughs in technology have helped the push towards 'mobility as a service', the benefits have long been recognised [27]. Two examples of early experiments of collaborative mobility are presented next.

2.1.1. Purdue University—Mobility Enterprise (1983)

In January 1983, Purdue University began its Mobility Enterprise, which provided families who joined the scheme with a MAV (minimum attribute vehicle or 'small car') and access to a shared pool of larger vehicles. The aim of the project was to use the availability of shared cars to enable drivers to choose a fit-for-purpose vehicle for each trip, rather than having to use an all-purpose vehicle for every occasion. It was expected that this would increase the efficiency of the vehicle kilometres travelled and reduce fuel consumption. The experiment ran for two and a half years and incorporated an average of 12 households. To encourage people to join their experiment, the Mobility Enterprise had to be competitive when compared with private car ownership. The results from this initiative showed a significant reduction in fuel consumption, but no change in the driving habits of those involved. The researchers found that they were not able to influence the driving behaviour significantly of those involved. However, the project was able to demonstrate that fuel consumption could be reduced through the use of fit-for-purpose mobility and that a tangible monetary saving can be used to motivate change [27].

2.1.2. San Francisco—STAR Project (1983–1985)

The STAR (short-term auto rental) project was undertaken in the San Francisco area between 1983 and 1985. The project identified an apartment complex of 9000 residents that was well serviced by public transport, and established a car rental dealership within the precinct, which provided competitively priced short-term rental agreements. The aim of the project was to demonstrate that it was possible to live in an American City without owning a motor vehicle. The study found that since the public transport services were well utilised during the commute to and from work, many of the trips made by private vehicle were discretionary in nature. For these discretionary trips, the presence of a rental dealership would allow drivers the choice of a fit-for-purpose vehicle, providing fuel savings and increased vehicle utilisation. The project was a partial success, with those who made infrequent trips electing to forgo private vehicle ownership (and in some cases saving as much as \$1000 a year). However, the project was limited in that it was only utilised for discretionary, non-work-related trips.

The project concluded that the scheme was well-suited to provide fit-for-purpose mobility in support of high-quality public transport [27].

2.2. Case Studies—Mapping the Value Beyond the Hype

The ability to reduce congestion, cut travel times, improve trip efficiency, save petrol, and reduce pollution through the use of collaborative mobility has long been understood [27]. The addition of high funding levels and the potential to create an entirely new consumer market within the transport sector offers a tempting reward for anyone who can unlock the potential of collaborative mobility [3]. With the capacity for profit, combined with environmental and social benefits, a wide range of international initiatives has been launched, with varying degrees of success, in an attempt to harness its potential [28]. The following section will analyse some of the key collaborative mobility projects that have been undertaken around the world. It will focus on identifying success and failure factors and will try to identify lessons that may be learnt from each case.

In recent years, the convergence of technology and infrastructure has renewed interest in shared mobility. Some of the emerging trends in this space are covered next.

2.2.1. Car Sharing Services

As most private vehicles are only utilised less than 10% of the time, shared car services, such as Zip Car and Autolib, aim to increase vehicle utilisation and efficiency by maintaining a fleet of vehicles that can be accessed on either a subscription or hire basis [29]. Global car sharing provider Zip Car estimates that due to the increased utilisation of the vehicles, each of the shared vehicles can replace as many as 15 ordinary vehicles. Although it should be noted that most Zip Car users drive between 60% to 80% less than the average motorist and this allows each shared car to provide for a greater number of people that would otherwise be the case [30]. Surveys of the users of Autolib, the French car sharing operator, revealed that of those who did not own a private vehicle, 70% listed Autolib as the reason that they had been able to move beyond private car ownership [30]. In recent years, some shared car providers have been developing new additions to their services and forming new partnerships to further realise the potential of collaborative mobility [31]. In an attempt to increase the attractiveness of car sharing in inner-city environments, companies are attempting to provide services that facilitate short-term, impulse use of shared cars. BMW's Drive now program uses flexible hire services to accommodate short distance trips. Proving initially to be successful, the scheme has now been established in five German cities, as well as London and San Francisco [30]. Ford's city driving on-demand scheme follows a similar approach and utilises a pay-by-the-minute hire scheme with the option for one way trips [32]. In Germany, the car-sharing operator Flinkster reached an agreement with Ford to have access to Ford's existing network of car dealerships as depots for the car-share scheme. In doing so, Ford and Flinkster have created a nation-wide car-share network without the capital for purchasing land and storage facilities [32].

Two other schemes currently being trialled by Ford are for non-corporately controlled car sharing. The "Car Swap" register is an internal experiment where a registry has been compiled of Fords employees in the Dearborn area in the USA. The register includes people who are interested in car sharing their vehicles. If someone who is part of the scheme needs a vehicle with some particular feature, a search of the registry would provide a list of people who can contact to acquire the use of that vehicle. Ford's second user-managed scheme, "Share Car", is currently in the development stage, with a team in Bangalore partnering with Zoomcar to establish the legal framework that would allow a group of people to jointly share ownership of a private vehicle [32]. A final scheme from the Ford motor group is the Remote Positioning project being undertaken in Atlanta, USA. The project is aiming to perfect the use of drone vehicles that can be controlled from a centralised facility, as shared vehicles. The use of these vehicles would allow a shared car operator to both deliver a vehicle to the customer's door and retrieve used vehicles from any point without the cost and time involved in physically ferrying valet drivers backwards and forth between jobs [32].

2.2.2. E-Hailing

The largest, most heavily-funded and most well-known on-demand mobility providers are the e-hailing services such as Uber, Lyft, and Didi [30,33]. Using private drivers who supply their own vehicles (to minimise the capital cost associated with traditional fleet operations), these companies operate an on-demand taxi-like service in most major cities around the world [29,30]. In particular, Uber is rapidly destabilising the traditional taxi industry [30,33] with recent reports showing that together with competitor ridesharing companies, they have captured more than 50% of the taxi market share for trips. Uber also has a ridesharing variant of its service known as UberPool [34] which operates in a number of cities including London. Statistics released by Uber claimed that, after six months of UberPool being introduced to London, the system has saved 700,000 vehicle miles (1,120,000 km) or 52,000 litres of fuel [35]. This saving translates to savings of 124 tons of CO₂ [34]. Uber also claimed that in some cities, over half of the trips are now being made using UberPool [35].

2.2.3. Public Transport Innovations

While most cities are continuing to push ahead with the expansion of existing conventional public transport networks (Beijing alone constructed over 230 miles (380 km) of subway between 2008 and 2015), the world is increasingly seeking ways to diversify the current transport options [29]. Active transport (walking or cycling) has recently been gaining acknowledgement as a legitimate transport option, with bike share schemes flourishing around the world [29]. In Paris, these two trends have been combined, with the city including its bike share service under the umbrella of its public transport network [29]. Probably the most innovative approach to public transport in recent times is the on-demand buses such as Kutsuplus and Bridj described next.

Kutsuplus—On-Demand Public Transport

The city of Helsinki in Finland has been praised for its 'ambitious' plans and target to overcome the need for private car ownership by 2025. The Kutsuplus mobility experiment initiated is seen by many as the flagship project leading this attempt [36]. Hailed as the first true on-demand public transport service, the Helsinki Regional Transport Service ran a technology-driven minibus service which utilised advanced algorithms to assign vehicles based on passenger demand on a real-time basis [28]. The initiative entered active service in 2012 with a fleet of three dedicated minibuses. Kutsuplus allowed commuters to specify an origin and a destination point (within a defined service area), and the algorithm then identified a minibus travelling in that direction and instructed its driver to pick up the new passenger [37].

The Kutsuplus experiment was launched with two initial goals: (1) test the technological feasibility of using computer-based routing algorithms to overcome the difficulty of maintaining effective control over mobility, and (2) measure public support and willingness to pay for on-demand public transport [28]. The service received strong public support with positive feedback and continual growth in ridership figures [28]. A survey of user satisfaction recorded an overall satisfaction rating of 4.7 out of 5, 10% higher than the satisfaction rating received by Helsinki's conventional public transport networks [37]. An important finding from this experiment, which is relevant to other collaborative mobility operators, is that people were willing to accept longer travel times during a journey more readily than they would accept longer waiting times at the start of a journey. This finding helped the operators to revise the algorithm used for bus routing to minimise waiting times on pickup which resulted in a considerable increase in the number of users [37].

Although the cost per trip was reported to be lower than that seen in conventional bus services [37], the Kutsuplus bus service was deemed a financial strain on the public purse and was discontinued at the end of 2015 [38]. This was contrary to public expectations of seeing the service expanded into the future [28]. To increase ridership, the system needed to provide a more attractive service covering a larger service area, longer operating hours, and shorter waiting times at pick-up. While some improvements in service quality were achieved, large-scale increases in the services required the

purchase of extra vehicles [37]. Data collected during the trial showed that every time the service was expanded, the increase in passenger numbers improved the efficiency of the entire system because a higher density of users allowed the computer algorithm to plot a more optimised route with less meandering to pick up lone passengers [37]. To that end, it was reported that a 31% increase in the system's capacity would result in an average of a 60% increase in income [38].

Other Examples of On-Demand Public Transport

The near-success story of Helsinki's on-demand minibus service has drawn interest from around the world [38]. Kutsuplus demonstrated to both governments and individuals that on-demand collaborative mobility was challenging but feasible from a technology perspective [37]. Since Kutsuplus' launch, cities around the world have begun to adopt similar on-demand shared mobility solutions. Many of these schemes are directly modelled after Kutsuplus, like Via Transportation and Chariot systems [29]. Other initiatives, however, have used Kutsuplus as a base framework but have then tailored the system to overcome identified weaknesses and meet the needs of the operator [38]. An example is a Bridj service, which includes a number of cities and project in the U.S. including Kansas City (Missouri) where Bridj collaborated with the city to provide a partial on-demand service. The city-owned Bridj busses follow a rough route in accordance with an established timetable (like in a traditional bus service) but will respond to the customer request for pick up along that route [38].

In both London and New York, Ford has launched Dynamic Social Shuttle, its own on-demand minibus scheme with a focus on minimising response time. Ford aims to use this project to gain data on shared mobility patterns and better understand the social dynamics that are driving shared mobility [32].

The Split project in Washington DC uses the same software as the Kutsuplus mini busses to optimise vehicle routes and travel times. However, it has succeeded in overcoming the growth difficulties that crippled its predecessor by using private vehicles for its fleet. This Uber-style approach where drivers provide their own vehicles reduces the financial barrier to growth which prevented Kutsuplus from achieving the scale of operations that would have allowed it to function effectively [32].

In Bangalore, rather than using dynamic pricing to affect the cost of bus fares throughout the day, the bus operators issue a raffle ticket to every person who travels on a bus outside peak hours, with cash prizes for the lucky winners. The scheme doubled the pre-peak ridership and reduced peak hour travel times by 24% at little cost to the bus operator [30]. As a point of comparison, Singapore's attempt to influence pre-peak ridership on their rail lines achieved a 7% shift in passenger numbers by making all travel before 8 a.m. free [30].

3. Social Impacts: Benefits and User Acceptance of FMoD Systems

In past decades, sharing a vehicle with unknown passengers was not popular. Today, there is a significant positive psychological change toward shared mobility. This has partly been encouraged through sharing economy models in mobility, which has facilitated new transport solutions such as car sharing, ride sharing, bike sharing, and ride sourcing [39–41].

3.1. Car-Sharing

In recent times, car sharing has been promoted as a viable alternative to car ownership. This is a model where customers become a part of a car-sharing program upon paying an annual subscription through which they will gain access to a fleet of vehicles. Users are charged based on the time they have used the vehicle away from the server pool. Moreover, gasoline, maintenance, and insurance fees are included in an hourly payment fee [42]. The car-sharing applications around the world can be categorised into three types: round-trip car-sharing; one-way car-sharing; and personal vehicle sharing. Car sharing's success includes operation schemes through 26 countries in 1100 cities worldwide and nearly millions of users in the past two decades only [43]. The first formal car-sharing companies have started in 1980 in Switzerland and Germany. Nowadays, car sharing in Switzerland has around 60,000

users in 900 locations with 2000 vehicles while there are about 75 companies with 40,000 members approximately in Germany. Recently, Car2Go started in Germany has expanded to 18 cities around the world with over 350,000 customers. This growth also reached the Netherlands, Austria, Sweden, and France. In 2008, the US, Europe, and Australia have reached 150,000 users of car sharing [44]. Car sharing is an integral part of the big picture 'shared mobility' vision, which enables access to services without owning them [45]. The two car sharing schemes, Business-to-consumer (B2C) and peer-to-peer (P2P) operate through smartphone and internet platforms. B2C is referred to as free-floating car-sharing, where users are not required to return the car to the pick-up location (i.e., a one-way trip is allowed). While P2C is where existing car owners rent their vehicles to others for a short period of time [41,46].

Many studies were conducted to study the effects of sharing a car, owned vehicle usage and the distance travelled. In a study in Bremen and Belgium [47], the authors found car-sharing systems have replaced 7–10 private cars (in Bremen) and 4–6 cars (in Belgium). In another study [48], it was shown that 30% of users have sold a car or delayed purchasing a new one in San Francisco, USA. Similar to Chicago, where the I-Go Car Share system has also decreased the purchase of private cars [49]. In Switzerland [50], there were reductions of 33–50% in car kilometres travelled and an increase in public transport usage after joining the car-sharing schemes. In Germany, car sharing reduced the number of private vehicle users by approximately 50% while in Ireland, public transport and alternative travel modes (such as bicycles, trains, buses, and walking) increased due to car sharing systems.

The advantages of car sharing compared to car rental services and private car ownership was also discussed in [47]. In comparison to car rental services, car-sharing companies attract users by offering features for more suitable access to the car such as convenient location and flexibility. Compared to vehicle ownership, care sharing is comparatively easy to use and its main goal is to replace private vehicle ownership and reduce the number of privately-owned cars. It is considered a good alternative for owning a car, particularly for people who drive less.

Surveys have also shown key considerations why people are encouraged to use car sharing. For example, in Singapore, these reasons included cost savings, flexibility in managing family trips without investing in a second car, or financial inability to own and maintain a car [51]. The same study discussed the car-sharing user satisfaction, where it was found that 76% of users are satisfied with the car sharing in Toh Yi and 50% in Bishan. In addition, the main features that increased the level of user satisfaction are the availability of the car, possibility to choose the desired car from a number of available options, convenient booking and payment arrangements, and the good condition of the car [51]. It was also shown [52] that the main reasons for using car sharing schemes in the UK include cost-effectiveness, reducing emissions foot-print, and users believing that they are making a difference by sharing.

3.2. Bike-Sharing

Bike sharing is the shared use of a bicycle fleet, which is delivered by a commercial provider. The principle behind this concept is that individuals use bicycles based on an 'as-needed' basis without having the responsibilities of ownership and recurrent maintenance and operations costs [53]. Bike sharing has gained popularity in recent years with an increasing appetite for sustainable modes of transport. It was also facilitated by increasing ease of use through a mobile app and their value in the first and last mile connection to other modes of transit [54]. Key features of bike sharing schemes include [55]:

- 1. Bikes can be rented from a particular location and could be returned either to the same place or to a different location.
- 2. Easy to access with diverse business models underpinned by the use of technology such as mobile phones and smart cards.
- 3. Provides first and last kilometre solutions that can be integrated with public transport systems

The largest bike-sharing scheme was launched in Paris with more than 20,000 bicycles in 2007. Today, China has the largest bike sharing scheme with over 70,000 bikes in Wuhan and 65,000 in Hangzhou [56]. New York City had a bike-sharing program with 10,000 bikes in 2013. Bike sharing is more convenient than the private bicycle as shown in a study conducted by [57]. The study shows that 60–70% of individuals in China believed that bike sharing is a more convenient option than using private bicycles. Similar results were found in Washington, DC, Minneapolis, and Melbourne [58,59]. Another study into bike sharing in North America was carried out by [59]. The study assessed bike sharing from both the operator and user point of view. The focus was on the impacts of bike sharing on other transport modes, user opinions and preferences, and the influence of travel distances on using bike sharing. The results showed that bike sharing has reduced walking, taxi, and car usage.

Currently, bike-sharing systems are used in more than 15 countries in over 75 cities around the world using more than 70,000 bikes. Major countries with a bike sharing system are Australia, Canada, China, France, UK, and the USA. Out of these, Europe's largest and most successful system, Velib, is in Paris [53,55]. Also, in Wuhan and Hangzhou, China operates the world's largest public bike-share schemes with over 65,000 bikes [60]. In addition, an online "capital bike-share customer use and satisfaction survey" was carried out in Washington DC in 2012. The results showed that 7% of 5464 of respondents were willing to shift from car usage to bike sharing [61]. Another survey in London found that 60% of the respondents were willing to shift to bike sharing [62]. There are many recent studies by Elliot Fishman on bike-sharing systems, Table 1 summarises the impacts of those relevant studies [63–67]. Other issues impacting their uptake include topography and climate in certain cities which do not suit bike sharing [68,69].

Study	Data	Methods	Results	Recommendations
	Data collected from a Hospital to assess bike share and non-bike share cities in USA	Statistical approach Statistical approach Statistical approach Statistical approach Statistical approach Statistical approach Statistical approach		 A reliable methodology/tool Is required to collect data
Risk Associated with Bike Sharing	Data for severe injuries collected from North American and European cities	Statistical Packages for the Social Sciences (SPSS), a Chi-square test and Incidence Rate Ratio (IRR) based on Poisson-regression with generalised linear models in SPSS	Low risk of fatal and severe injuries of a cyclist after a comparison between general and private biking schemes.	 and reports related to the safety of Bike sharing schemes Bike sharing should be introduced with safety measures ensured by decision makers such as traffic calming
Safety of Electric Biking and Classic Biking	Data on crashes collected from victims treated at emergency departments (ED) and information on cyclists for electric bike (EB) and classic bike (CB) users In Netherlands	Binary logistic regression	 EB users are involved in single biking incident while CB users are more likely involved in collision with other road users. Frequent bike users (4–7 days a week) aged 50–69 and CB users are more often treated at ED than other users. 	More research needs to be conducted to study the impact of electric biking on road safety.
Factors Affecting Usage of Bike Sharing Schemes	Online survey for Bike sharing members and non-members in Melbourne and Brisbane	Logistic regression model	 Accessibility of bikes. Provide more stations near the workplace especially for employees under 35 years old. Increase bike lanes to increase safety will increase membership. Users with membership have a higher income than non-membership users since stations are located in the city area, not the suburb. 	More studies are required on how to attract more users (in terms of all income levels) to become members in bike sharing programs Further research on the impact of compulsory helmet legislation on the usage of bike sharing.

Table 1. Recent studies and their impacts on bike-sharing systems [63–	67	7]
--	----	---	---

Study	Data	Methods	Results	Recommendations	
Impact of Public Transportation System on Bike Sharing Mobility Pattern	Bike sharing trips in London	Time series spatial and temporal analysis of bicycle trip counts and durations	 85% increase in bicycle trip counts and 88% increase in trip duration because of public 	 85% increase in bicycle trip counts and 88% increase in trip duration because of public Quantitative and qualidata needs to be collec public transport and b sharing users and mor 	Quantitative and qualitative data needs to be collected on public transport and bike sharing users and more
		Complex network analysis: in which each bike station is a node and a link is formed between two stations	 transport systems distributions Improve connectivity of network and enhance interactions between biking stations. Bike sharing can reduce reliance on public transport during disrupting events. 	research is required to analyse their behaviour. Interactions between public transportation systems and bike sharing schemes are important and should continue to include integration of ticket systems and locations of bikes' stations. Hence, it will help to plan mobility options ahead during disruptive events.	

Table 1. Cont.

3.3. Mobility-as-a-Service (MaaS)

The drive towards collaborative mobility has received strong support, with the transport sector receiving the highest levels of funding for any sector within the collaborative economy [70]. This is manifested by one of the most promising trends within the disruptive mobility space, known as 'Mobility-as-a-Service' or (MaaS). The key concept behind MaaS is to place the road users at the core of transport services, offering them mobility solutions based on their individual needs. This can be achieved by providing a single platform for combining all mobility options and presenting them to the customer in a simple and integrated manner. This means that easy access to the most appropriate transport service will be included in a bundle of flexible travel options for customers. MaaS has the potential to fundamentally change the behaviour of people and reduce reliance on car ownership by providing easy on-demand access to the mobility services they need (Table 2). The trend is therefore gradually shifting from the provision of buses, trams and trains to a focus on what people require, and how a more considered and integrated approach could produce better outcomes [30].

	Traditional Mobility Solutions	Ne	ew Mobility Services
	Private car ownership	Car sharing: peer to peer	A peer-to-peer platform where individuals can rent out their private vehicles when not in use (e.g., Turo)
Individual-Based Mobility	Taxi	E-hailing	Process of ordering a car or taxi via the on-demand app. App matches the rider with driver and handles payment (e.g., Uber, Lyft, Didi)
	Rental cars	Car sharing: fleet operator	On-demand short-term car rentals with the vehicle owned and managed by a fleet operator (e.g., GoGet, Car2Go, ZipCar, Getaround
Group-Based Mobility	Car pooling	Shared e-hailing	Allows riders going in the same direction to share the car, thereby splitting the fare and lowering the cost (e.g., UberPool, LyftLine)
	Public transport	On-demand private shuttles	App and technology-enabled shuttle service. Cheaper than a taxi but more convenient than public transit (e.g., Bridi)

Table 2. New urban mobility services (adapted from [29]).

The key advantages to these new services include reliability, predictability, convenience, and ease of accessibility. Most services also offer easy and secure payment options using cashless mobile

transactions. For a given trip, consumers can select from different types of services based on trip distance, waiting and travel times and levels of service [71].

Another key feature of MaaS is that it combines the use of public and private transport solutions to provide near door-to-door service [72,73]. It uses different modes of transport to deliver a tailored mobility package including complementary services such as trip planning, reservations, and payments through a simple mobile application [74]. A single interface is provided through which customers create and manage their total journey, with payment to all service providers coming from a single account and a single payment. Therefore, the main concept behind MaaS is to provide mobility solutions to travellers [75,76] based on their requirements. The scenarios for future urban mobility and the way to get there are schematically shown in Figure 1.



Figure 1. Digital-age urban mobility options (adapted from [75]).

The MaaS model offers opportunities to enhance public transport to cater to customer expectations through a connected and integrated transport system [77]. A connected transport system integrates transport modes with the overarching goal of providing unified door-to-door service to the passengers. In the past few years, MaaS initiatives grew rapidly especially in Europe, and some of these existing travel services or initiatives offered to customers through monthly subscription schemes include MOBIB in Brussels, Belgium [78], HANNOVER Mobil [79], EMMA in Montpellier, and SMILE in Vienna [80]. When implementing the MaaS concept, it is important to identify the factors that are

important to set up a reliable, working and successful scheme. Some of the core characteristics of MaaS implementation [19,81–86] are listed in Table 3.

	Characteristics	Description
1	Integration of transport modes	This allows bringing together multi-modal transport systems to allow users to choose and facilitate their intermodal trips. Such as public transport, taxis, car/bike/ridesharing, and on-demand bus services.
2	Tariff option	MaaS would allow users to choose from two options, "Mobility package," and "Pay-as-you-go" to pay the tariff. These would be modelled based on kilometres travelled, minutes spent, and points collected on travels.
3	One platform	MaaS relies on a digital platform that allows users to do trip planning, booking, payments, ticketing and acquire other necessary information such as weather details.
4	Multiple actors	The system is built upon the interaction between different parties through a digital platform. They are end-users of mobility services, service suppliers, platform owners, and external stakeholders like local authorities, telecommunication, data management companies, and payment clearing agencies.
5	Use of technologies	MaaS is enabled through the integration of technologies such as Wi-Fi, 3G, 4G, LTE, and GPS, e-ticketing, e-payment, Internet-of-Things (IoT) and database management systems through mobile devices or computers.
6	Demand orientation	Best possible transport solution is offered based on the perspective of the customer via a multimodal trip-planning feature.
7	Registration requirement	Customers need to have registered with the system to access the available services. This either could be a personal account or for an entire household. The subscription also allows users to personalise their travel demand profiles and preferences.
8	Personalisation	Users requirements are made more efficient based on the uniqueness of different customers. The system would provide tailor-made and specific solutions to the users based on their profile, preferences, and history records.
9	Customisation	This enables customers to change/modify the offered services according to their needs. This is a key characteristic and attractive feature of MaaS.

86]
-86

Furthermore, MaaS can include other attributes such as gamification where customers are rewarded for taking environmentally friendly trips; access to parking, park and ride services; and subscription plans based on customers' monthly travel budgets [87–89]. Parking pricing can come very handy to encourage Maas in a more effective and efficient way [90].

MaaS solutions benefit users in numerous other ways, especially providing customers with new flexible transport modes, such as point-to-point (taxi + public transport) or point-via-point-to-point (ride sharing + public transport). Increasingly, these flexible transport solutions could pose a threat to conventional public transport, particularly suburban buses.

To enable personalised services with seamless trip chains, it is important to have integrated transport modes. The digital platform should also provide options to pre-book and pay for a trip package through a mobile application.

Another important aspect to highlight is that MaaS solutions provide customers with transparent costing of trips. Many consumers do not have a complete understanding of the actual costs of travel

in private vehicles. However, with above-mentioned mobility services including car sharing, users can compare the traveling cost and determine which mode is more economical to use. For example, a case study in Finland conducted in 2011 showed that car-related travel accounted for 80% of the total cost of transport per household [91]. Flexible choices such as car sharing and on-demand ride-sharing services would be an effective replacement for private cars. The study also found that for shorter travel distances, walking or cycling would be more effective.

3.4. Environmental Impacts

Car dependency is a major contributor to environmental emissions in cities [11–13]. Melbourne is one example where 75% of the trips are made by car [92]. Passenger vehicles account for consuming 18 million barrels of oil each day which result in 2.7 billion tons of carbon dioxide annually, worldwide [24]. Low occupancy cars also contribute to traffic congestion resulting in many detrimental effects such as increased fuel consumption, higher risks of accidents, more vehicular emissions, and increased transport costs [93]. In Melbourne, urban traffic congestion accounted for approximately \$3 Billion in 2005 and is predicted to double by 2020 [94].

Therefore, public transport services are essential to address the above-mentioned environmental and cost issues. Shifting passengers from low-occupancy vehicles such as single-passenger private vehicles to shared transport is one of the key strategies pursued by city decision makers around the globe. Car-pooling, for example, is a powerful strategy for delivering passenger trips that have similar origins and destinations. This is beneficial for both users and the wider community and can essentially be considered as a form of public transport. At the individual level, users will have a reduction in the cost of petrol and maintenance of their cars while the city and community will have reduced traffic congestion and pollution [95]. In a study conducted to identify the economic effects of a free-floating car sharing system in Germany [96], the authors analysed data from the car2go car-sharing scheme. This scheme allows users to take and leave vehicles at any point within the city and does not require prior booking and users are charged a fixed $0.19 \notin/min (0.25\%/min)$ for their journeys. The survey indicated a reduction in CO₂ per average car2go user and a contribution to reducing private vehicle usage in the cities. In another study for Singapore [51], the authors showed that car sharing is more affordable for customers who are charged based on time and distance. The car-sharing scheme was also found to have contributed to reducing traffic volumes.

Furthermore, [96] noticed a major change in emissions as a result of the Car2go system in Germany. The study looked at the lifecycle of vehicles starting with manufacturing, raw material consumption and energy usage. It also looked at the operation of the system including emissions and fuel consumption. Emission results included a reduction of total emissions CO, CO₂, NOx, and SOx with a further reduction in traffic-related noise. In addition, [97] further stated that there could be a 5–8% reduction in fuel consumption with the implementation of ride sharing.

Autonomous vehicles are also being promoted as a form of future public transport. A number of studies have examined the advantages of the use of shared autonomous vehicles (SAVs) including environmental benefits and cost efficiency. One study [98] found that total kilometres of personal car travel would be reduced leading to a reduction in private car purchases. In another study [99], it was estimated that SAVs are more cost-efficient as they are expected to reduce trip costs by one to eight dollars per mile (on average) due to automation and removal of human labour. Another study [100] which looked at the impacts of SAVs on environmental sustainability, the authors found that each SAV has the potential to replace approximately ten privately owned vehicles. They also predicted a reduction in energy consumption, greenhouse gas (GHG) emissions, and air pollutants emissions. Thus, evidence from multiple studies suggests that car sharing (current and emerging forms) is environmentally feasible and economical.

Therefore, the implementation of a flexible on-demand public transport system, which encourages people to minimise the use of low occupancy vehicles, would be environmentally feasible and cost-effective in the long run.

4. Technology Trends and Developments—Underlying Disruptive Technologies in FMoD

The success of FMoD systems is due to a number of converging factors including the fast pace of development of disruptive technologies, which are facilitating the wide adoption of these mobility solutions. Some of these key technologies are described next.

The shift towards embracing collaborative mobility is driven by the recognition of the potential opportunities, and risks generated as a result of the interaction between existing and emerging disruptive technologies [3]. The use of collaborative mobility has been seen as a way of harnessing new developments, like big data analytics and the Internet of Things, to mitigate against the effects of potentially negative trends, like increased congestion and pollution resulting from continued urbanisation [27,101].

With population growth expected to add an additional billion people to the world's inhabitants within the next decade, and with 60% of these people expected to live in cities, transport operators are faced with a situation where traditional methods will no longer be sufficient to restrain congestion [29,102]. Under these conditions, increasingly large amounts of effort are being diverted into alternate means of providing mobility [33]. In addition to the efforts by transport authorities, who are already struggling to keep pace with existing congestion levels, the predicted migration of the earth's population towards its cities will cause congestion to spread to regions that have never before experienced traffic issues [29,38]. The following section reviews few of the opportunities and trends and identifies ways that they have the potential to influence the drive toward collaborative mobility.

4.1. Digitalisation and the Internet of Things

Since the conception of collaborative mobility, attempts to organise large-scale programs have been hampered by the difficulty of establishing an effective central administration [27]. The underlying premise behind any collaborative mobility enterprise is to increase transportation efficiency by increasing vehicle utilisation. This cannot be easily achieved without strong centralised control of the vehicle fleet [30]. Early examples of failed attempts included the Procotip (Montpelier, France, 1971) and Witkar (Amsterdam, 1973) shared car schemes. Both of these schemes made compact small vehicles available for use in and around the congested inner-city areas using a payment token system. Both of these schemes failed because of inefficiencies in vehicle distribution caused by a lack of control over the vehicle fleet, resulting in poor asset utilisation and excessive downtime between each trip [27].

At the same time, the advent of the 'Internet of Things'—whereby people and physical objects are constantly interconnected to the Internet—has also exacerbated effective control of such undertaking [30]. Operators are now provided with unprecedented levels of data, including real time analysis on everything from vehicle positions to weather patterns. With this data available, the challenge facing transport decision makers is no longer about gathering information, but how to use the information available for decision making in a meaningful and timely manner [103]. In Finland, the Kutsuplus bus network received constant updates on traffic conditions, allowing drivers to provide updated estimates of arrival times to their passengers, often accurate to within 30 s [37]. Around the world, software developers have developed tools to combine information on every transport option available in a given city. Users simply enter their origin and destination, and the software would provide them with a route optimised for cost, travel time, comfort, or environmental efficiency [36].

4.2. Mobile Computing

The 2018 global digital report reports the number of mobile phone users to have reached around 5.125 billion users, and 4.021 billion internet users, worldwide. People tend to use internet and mobile phones with GPS systems to get help with the navigation to their desired destination and to get information about real-time information about traffic and updates about the shortest or fastest routes. Mobile computing together with wireless communication and remote sensing are considered the key driving forces behind today's Intelligent Transport Systems (ITS) which are creating a huge leap

forward in future mobility. Vehicles embedded with sophisticated computer systems and on-board sensors have also allowed for collecting information about the environment, the location of the vehicle, and to exchange information with other nearby vehicles in real-time [104]. This has enabled transport systems operators to undertake a real-time analysis of travel patterns and establish adaptive management to operate transport systems more efficiently.

4.3. Internet of Things (IoT)

The use of IoT facilitates reliable IT-based infrastructure to provide an interconnection through of computing devices, which enables to exchange data in a secure manner. This has been possible due to the use of the internet of computing devices that are embedded into everyday objects that are allowed to be tracked, coordinated, and controlled across the internet or across a data network along with the use of sensors and actuators [105]. Currently, the use of IoT in transport is growing, as there are many potential areas of applications, such as developing a cloud platform to access a vehicular data, which enables the exchange of transport-related information [22]. Especially, information like vehicle location tracking and monitoring, road conditions and maintenance, and incident information, which are useful for traffic control and management. However, there are challenges to overcome with the developments of transport services through IoT, and the challenges are mainly in terms of security, privacy, scalability, reliability, lack of global standards, and service quality [22].

4.4. Cloud Computing and Fog Computing

Cloud computing can be categorised into public and private clouds. Public clouds include services offered by commercial providers such as Google App Engine and Amazon Elastic Compute Cloud. Private clouds are usually reserved services that are provided to specific end-users [106].

There are three main services enabled due to cloud computing including:

- 1. Software-as-a-Service (SaaS): The end-users are allowed to execute their applications remotely.
- 2. Infrastructure-as-a-Service (IaaS): The end-users are able to access the physical computing resources (computers, virtual machines).
- 3. Platform-as-a-Service (PaaS): Users are provided with tools and libraries by the cloud, to develop their own software.

In transport, advances in cloud computing and IoT are providing opportunities for better traffic management and customer-oriented services. As an example, to improve vehicle-to-vehicle communication and road safety, a novel vehicle cloud architecture named ITS-Cloud has been proposed [106]. Moreover, to optimise traffic control, cloud computing has been used to develop a cloud-based urban traffic control system [107]. However, cloud computing faces a number of challenges in the transport context, including cybersecurity and also establishing unified architectural frameworks with convenient functionality [106].

Cisco put forward a novel paradigm, called fog computing, which extends cloud-computing services to the edge of the network, and enables the communication, computation, and storage closer to edge devices and end-users. The benefits of fog computing include enhancing low-latency network connections between devices and analytics endpoints, reducing network bandwidth, fast data processing, security, and privacy [108]. Advances of fog computing mainly enable connected cars, smart cities and real-time analytics [109]. Reference [110] discuss the challenges that the existing traffic light control systems are facing, such as avoiding heavy roadside sensors, resisting malicious vehicles and avoiding single-point failure and how they have used fog computing to propose secure intelligent traffic light control schemes to overcome above-mentioned challenges. However, fog computing is challenged by security issues such as man-in-the-middle attack (MITM), i.e., an attacker actively eavesdropping the direct communication between two parties [111].

4.5. Crowd-Sourcing and Data Fusion

One of the key challenges facing mobility providers and data analysts is how to combine and fuse data streams from separate, often incompatible sources and combine them into a meaningful, user-friendly and cohesive set of information [112]. The computer algorithm used to direct the Kutsuplus busses was one of the first to combine user requests with spatial positioning and real-time traffic reports to control route planning and predict accurate estimates of arrival times [37]. The success of this computer algorithm has paved the way for a new era of informed journey planning [38]. By combining multiple information sources into one interface, tools like London's "CityMapper" application (now expanded to cover 10 EU cities) can increase the efficiency of people's mobility by allowing them to view all of their options in real time, and optimise their journeys [30,112]. Other initiatives such as Moovit and Moovel are also being used around the world with each system offering some unique features [30,113,114]. For example, the Moovit App augments its data collection by utilising customer feedback to increase the accuracy of its route recommendations [30]. The Waze App, while not a journey planning app, advises drivers of traffic congestion and disruption in real time, guiding their vehicles around hotspots to reduce travel time and save fue [29].

4.6. Data Analysis Techniques

This section of the paper presents some of the tools that have been applied in FMoD applications and case studies. The discussion in the following sections will highlight the challenges and limitations of these approaches and outline directions for future research.

4.6.1. Linear Regression

Linear regression is a statistical analysis technique used to model the relationship between dependent and independent variables [115]. Regression techniques are well-known and have been used successfully to uncover information from dense transport data [116]. This includes recent studies where this technique has been used for data analysis as related to FMoD. For example, the effects of CO_2 emission changes due to the adaption of car sharing were investigated in [117]. First, mixed logit models were used to determine the preferences of using car sharing, and then binary logit models were used to determine if individuals are willing to shift from using their private cars to using car-sharing. Then, a linear regression model was used to analyse the behaviour of individuals and their social, economic, and environmental motives behind switching from using a private mode of transport to car sharing. Similarly, the factors that encourage private vehicle owners to shift to shared mobility were analysed using multilevel regression models [118]. Regression models have also been used in related applications to predict the travel times for different routes for public transport. Simulated data were used to predict average travel time for a bus route and overall bus travel times [119,120]. The independent variables considered for regression analysis included the length of the route, speed, frequency, flow intensity per Km, and the number of passengers. The results showed that it was valid to use regression models, but the models needed to be validated using field data.

Linear regression models were also used to predict the flow and demand of bike-sharing in Lyon [121]. In Greece, they were used to predict travel demand based on multiple linear regression analysis [122]. In another study, Harbin, the capital city of Heilongjiang province was selected to test the factors affecting people's choice of using customised bus service by using logistic regression (predictive analysis) [123]. Furthermore, another study used linear regression models to determine the number of passengers by usage and operational aspect in Great Britain [124]. In general, regression analysis has been used extensively in the past as a tool for data analysis, although in recent times there has been a noticeable shift towards applications of AI for data analysis which is being facilitated by fast computation techniques that can be used to identify patterns among variables [125].

4.6.2. Time Series Analysis

Data sets, which include time-ordered sequences of observations, are defined as a time series. Generally, time series observations are correlated and most standard statistical methods based on random samples are not applicable. Instead, time series analysis is used. These include two approaches [126]: time domain and frequency domain approaches. In the time domain approach, an autocorrelation function (ACF) and partial correlation function (PACF) are used as time functions to describe the characteristics of a time series with an evolution represented with various time-lag relationships [127]. One of the most widely uses of time series analysis is to forecast future values.

For example, time series analysis was used to evaluate human mobility [128] using the number of available bikes in stations in Barcelona. After detecting temporal and geographical mobility patterns within the city, time series analysis was used to predict the number of available bikes for any station. Another study [129] showed how monthly time-series data could be used to explain aggregate demand for public transit in particular areas based on the prices of private and public transport, service characteristics, comfort levels, etc. The study was completed for Montreal and the results showed that the model could be used to predict the monthly number of adults likely to use the Montreal Urban Community Transit Commission (MUCTC) services. In another study, the temporal aspects of the relationship between petrol prices and public transport in US cities were analysed for the period 2002–2009 [130]. Time-series analysis was used to estimate any presence of lagged effects of price and service on transit patronage.

4.6.3. Clustering and Classification Methods

Data clustering is an unsupervised classification of patterns in a dataset. In transport applications, the clustering could include classifying the data into groups and creating general representations that are useful to optimise public transport services [131,132]. For example, access to smart card data from automated fare collection systems has been used to analyse passenger behaviour and resulted in classifications of groups of passengers that depict similar behaviours [133]. Clustering methods fall into several categories such as space partitioning, hierarchical methods, and density-based methods. One of the most commonly used space partitioning methods is the k-means algorithm, which is used to cluster n number of objects into k partitions where k < n. In a study to predict online bus arrival time using k-means cluster algorithm [134], the authors introduced a method by developing delay data clusters in accordance with the delay and time of the day. Next, hierarchical agglomerative clustering (HAC) was used to develop a hierarchy tree, by progressively merging clusters from the individual elements. The HAC is used to study passenger weekday profiles in order to uncover patterns in passenger travel behaviour [135,136]. The DBSCAN is a commonly used density-based algorithm, which is capable of identifying the common groups in a large spatial dataset through observing the local density of corresponding elements [137]. Moreover, k-means and DBSCAN clustering methods have been used to identify passenger boarding and alighting times, and location by analysing the smart card data [138–142].

Classification techniques are useful to discern patterns in the data as well as predict a future outcome based on historical observations [143]. Some of the other well-known classification techniques include the naive Bayes classifier, decision trees (DT), artificial neural networks (ANN), and support vector machines (SVM) [144]. Probabilistic and k-nearest neighbour classification models have also been used to classify activity-based travel choice patterns, which are useful to cluster users based on their activity patterns [145]. AI tools can be used to extract important features and predict quantitative and qualitative transport data in real time [146]. Clustering and classification [147,148] analysis methods are used to uncover hidden patterns in streams of data. For example, clustering and classification methods have been used [149] to study train station crowd patterns with the use of smart card data. The authors used HAC and dynamic time warping (DTW) clustering methods to regroup stations based on passenger usage patterns and then introduced three classification techniques

to forecast the crowdedness of stations. These advanced data analysis methods have been shown to be useful for knowledge extraction when the data includes rich information on group behaviour.

4.6.4. Machine Learning

Machine learning can help to detect spatial and temporal features from transport data [150,151]. Spatial data are usually data identified within a location such as information on the public transport routing, stops, and road network. While temporal data represent a time state data that could include, weather information, traffic conditions, timetables etc. The algorithms used in MaaS, for example, include co-training, co-regularisation, and margin-consistency style algorithms. These are generally referred to as multi-view learning algorithms as noted by [152]. Also, machine learning can help to detect the relationship between cause and effect from variables and also to find a common behaviour that two variables share [153]. Machine learning has also been used to determine parameters that affect the development of transport networks around the city such as the behaviour of network users, peak hours, and incidents during the day. This variation can be recognised to ensure that the users of that network have safer and more convenient trips with fewer delays.

In another study [154], the authors tried to understand the behaviour of public transport users through their smartphones. The study used information from the Global Positioning System (GPS), Geographic Information System (GIS), and sensors on the smartphone and combined it with machine learning techniques to detect and classify people's mode of transport, i.e., walking, train, bus, and taxi. The authors used different machine learning techniques and achieved an accuracy of 95% and above [155,156]. AI and machine learning techniques have helped in launching different MaaS mobile applications, worldwide. For example, the Whim app in Finland obtains data on people's preferred mode of transport and suggests the best way to get to a destination by booking online through a different mode of transport. Moreover, Qixxit and Moovel in Germany, Beeline in Singapore, and Ubigo in Sweden share similar techniques [157–160].

4.6.5. Big Data Analytics

Big data analytics are used to analyse data that is characterised as being unstructured, vast, and fast moving which is difficult to manage using traditional methods. Big Data tools include key technologies like Hadoop, NoSQL, MongoDB, and HDFS. These tools are applied to extract valuable information from data [161,162]. These techniques have been applied successfully in marketing, fraud detection, risks quantification, automated decisions for real-time processes, better planning, and forecasting [163,164]

A recent study has explored smart cities applications of Big Data by proposing an analysis service based on cloud technology [165]. The authors suggest that these can be further developed to generate information intelligence and support decision-making in smart future cities [166]. The authors presented a business model of Big Data for smart cities, which they believe could be used as a benchmark for future development of smart cities.

4.6.6. Optimisation Techniques

Optimisation problems are often complex and solutions are not readily obtained using a direct approach [167,168]. A major contributing factor to the complexity of a problem at hand is the number of decision variables [167]. There are numbers of approaches to solving large-scale optimisation problems which have a large number of decision variables [169]. Some of the approaches include stochastic optimisation, robust optimisation, simulated annealing, and convex–concave regularisation [170–174]

Stochastic optimisation is a collection of methods to minimise or maximise an objective function when there is randomness. In the past few decades, these methods have been useful in applications related to many fields such as science, engineering, business, statistics, and computer science [175]. Four of the broadly used methods are as follows: (i) sample average approximation, which is a two-part method dealing with sampling and deterministic optimisation [176,177]; (ii) stochastic approximation,

which is an iterative method which uses noisy observations to find the root of a function [178]; (iii) response surfaces is a set of methods which fit a surface to a set of decision-response pairs and search the surface to derive a new decision [179,180]; and (iv) global meta-model optimisation, which relates to an expected output through a set of inputs through regression [181]. Stochastic optimisation is an area with much active research in transport. For example, [182] has used the stochastic quasi-Newton method to produce a destination choice model with pairwise district-level constants for trip distribution based on a nearly complete regional OD trip matrix.

Metaheuristic optimisation is related to optimisation problems using metaheuristic algorithms. These metaheuristic algorithms are generally nature-inspired. From simulated annealing to ant colony optimisation, and from particle swarm optimisation to cuckoo search, many new metaheuristic algorithms have developed in all areas of optimisation [183]. Metaheuristics are useful in optimising block-box systems, where no gradient information and explicit information are available [184].

A solution to an optimisation model is defined as 'solution robust' if it lays near optimal for all scenarios of input data, and 'model robust' if it lays almost optimal for all data scenarios. Such a model formulation is defined as robust optimisation [185]. Robust optimisation is generally used to plan large-scale systems that are subjected to noisy, incomplete or uncertain input data [186]. It is widely used in many real-world problem domains [185,187] Reference [188] proposed a novel and reliable bus route schedule design problem using robust optimisation. The authors accounted for the uncertainty of bus travel time and the bus drivers' schedule recovery efforts. They developed a robust optimisation model to minimise the product of a weighting value and the sum of the expected value of the random schedule deviation and its variability. They found that optimal scheduled travel time is dependent on bus drivers' schedule recovery behaviour, and decision makers' scheduling philosophies. Furthermore, [189] used a robust optimisation model to formulate and solve a bus transit network design problem. Through the robust optimisation, they aimed to minimise the product of a weighting value and the sum of the expected value of use of the sum of the expected value optimisation model to minimise the product of solve a bus transit network design problem. Through the robust optimisation, they aimed to minimise the product of a weighting value and the sum of the expected value of operator cost and its variability.

4.6.7. Deep Learning and Reinforcement Learning

Deep learning is a class of machine learning algorithms based on artificial neural networks. These are capable of learning the representations of data with multiple levels of abstraction. There are few types of deep learning architectures such as recurrent neural network (RNN) [190–193], convolutional neural network (CNN) [194–196], and deep belief net (DBN) [194]. Deep learning applications have grown rapidly in pattern recognition, signal processing, discrete choice modelling, and optimisation [197–201]. References [202–204] provide a good introduction to deep learning methods for learning of feature representation either supervised or unsupervised at more abstract and successively higher layers.

Transport-related problems such as, passenger flow prediction are complex and are usually non-linear problems that are influenced by many fixed and stochastic factors [205]. Recent advances such as powerful graphical processing units (GPUs) have enabled deep learning methods to exploit complicated, compositional, and non-linear functions effectively. These advances have also made it possible to learn hierarchical and distributed feature representations accurately and make effective use of labelled and unlabelled data [206]. Reference [206] developed a novel model for passenger flow prediction using deep learning methods. They used their model for Xiamen BRT stations to predict hourly passenger flow. The flow prediction model is combined with unsupervised training model based on stacked auto-encoder (SAE) and supervised training model based on deep neural network (DNN). Reference [207] also developed a deep-learning model to estimate bus passengers destinations with the use of entry-only smart card data and land-use characteristics.

Current research into solving public transport-related issues and developing mobility-on-demand services using deep learning methods are still at the early stage. The potential of reinforcement learning, which is a type of AI paradigm used to analyse interactions with the environment and learning from their mistakes, are very promising [208]. This can be used in route choice behaviour, which enables

the possibility of providing feedback on foregone or non-chosen alternatives [209]. Reinforcement learning is also an active research area in autonomous driving [210] and for predicting the demand for AMoD services.

5. Existing and Emerging Business Models

Vehicles in FMoD fleets can be categorised into a number of services as summarised in Table 4 and described below [19].

Model	Characteristics
Taxi services	High costMost conventionalHighest personalized service
BRIDJ	Uses real time traffic data and passenger inputsAbility to find fastest routes
UBER, LYFT, DiDi and Careem, Lift hero, Hop Skip Drive	 Shared service Available on demand Availability of digital platforms Less expensive Enhanced passenger safety and security
Van pooling (Uberpool and Lyftline)	 Available on demand Available through digital platforms Shared service Reduced travel cost
Optibus, Customized bus	PersonalizedFlexible and demand responsive

Table 4. Existing and emerging business models and characteristics.

5.1. Shared Vehicles

Shared vehicle fleets have flexible schedules and routes, operate on-demand and focus on delivering the door-to-door convenience. Taxi services are the most conventional form of such business models and provide the highest personalised service for users, but comes at a high cost compared to other services. In more recent times, this business model had been challenged by the pioneers of shared on-demand services, including Uber and Lyft. The success of these companies has been facilitated by digital platforms that help users connect with drivers who offer different levels of service (e.g., Uber Black versus UberX). These companies have also recently introduced car-pooling services where passengers can share their rides with other passengers travelling in the same direction. The shared transport applications also include car sharing and bike sharing as described in the previous section.

As mentioned before, a key success factor of ridesharing is the plethora of web and smartphone-based digital platforms and solutions related to the transport sector [211]. Ride sourcing was initially defined as a type of ridesharing which provides services with the use of GIS and GPS technologies on internet-enabled mobile phones to order or organise real-time ridesharing [212]. Ride-matching software is used to match riders to drivers, automatically, with similar trips and notifications sent via smart devices [212]. Unlike conventional taxi services, passengers do not need to speak over a phone or hail a cab or even have the correct amount of cash ready for payment. They simply book their rides through a mobile application. The digital platform then matches drivers with travellers and handles payments. Although the taxi industries have now developed e-hailing services through mobile apps, they still operate on a conventional taxi company model and are more expensive than their competitors [213]. Uber (offering different services such as UberX, UberXL, and UberSelect) [214], Lyft in North America [215], DiDi in China and Australia [216], and Careem in the

Middle East [217] are among the most known ride sourcing service providers operating in many cities around the world. Although there have been some concerns about passenger and driver safety in these services, these companies are increasingly scrutinising their drivers and providing new functionalities in their apps to improve the safety and security of passengers and drivers. Other examples of these business models include Lift Hero in San Francisco [218] which was established to serve elderly and disabled passengers. In addition, in the same city and Los Angeles, there is a Hop Skip Drive [219] to facilitate children with their school rides.

Other applications include van ride splitting or pooling. These are interesting transport demand management tools which have the potential to reduce single-occupant vehicle trips [220]. It is now an emerging alternative mode of transport for commuters, particularly that they are available through digital platforms [221,222]. With this, the users get the opportunity to share the ride with other travellers whose trips are on similar routes and split the fare among each other; therefore, the travel cost per passenger is reduced. Services like UberPool and LyftLine have taken the lead to provide the service of ride splitting or pooling. Among the strategy that LyftLine uses is that they encourage the passengers to gather in a specific location in the city, and passengers are offered a discounted fare if they walk to the pick-up locations. Similarly, UberPool introduced smart routes with a similar strategy.

Demand-responsive shared transport (DRST), a service which is provided through a fleet of vehicles, booked via a mobile application by the users, and scheduled in real-time to pick up and drop passengers based on their needs. This service is a blend of individual door-to-door ride services and a typical transit system. Here, the operators' intention is to select an optimal approach to assigning vehicles to passenger's needs. From the users' perspective, there is a need to reduce additional travel time and distances that they have to experience due to shared rides, and possible reduced fees for individual rides. Recently, the efficiency of schedules of dynamic demand responsive transport (DRT) services are studied through simulation models. In this regard, agent-based models (ABM) were found to be effective in reproducing complex social systems and overcome certain limitations [223–225].

5.2. Flexible On-Demand Mini-Buses

Flexible on-demand mini-buses operate to deliver door-to-door convenience at a fraction of the cost of ordering a taxi. In recent years, there were many attempts and trials around the world to improve conventional bus services and provide convenient services to users through such business models. Examples include Optibus, which launched an AI-driven on-time optimisation solution [226]. China has also introduced a personalised and flexible demand responsive public transport service called customised bus (CB) [227,228].

One of the well-known on-demand public transport services is BRIDJ which was launched in 2014 in Boston, Massachusetts [229]. The service was subsequently expanded to Washington DC and Kansas City. This service uses real-time traffic data and passenger inputs to establish origin-destination data. The digital platform is capable of finding the fastest route and only stops at locations requested by a passenger to optimise the service. However, BRIDJ operation came to a halt in the US due to low ridership, the operation only during certain hours, lack of public awareness or marketing, and network-specific issues. Another application of on-demand minibus services is the Kutsuplus in Helsinki [230] which also came to end after a few years of operation, due to similar reasons.

It should be mentioned here that BRIDJ has been acquired by an Australian-based transit company and the system is currently being trialled in Sydney [231]. The key focus of the existing trial is to explore the feasibility of introducing a flexible, well-coordinated system operation as a reliable means of 'first and last kilometre' travel solution.

5.3. Autonomous Mobility-on-Demand (AMoD)

Autonomous mobility-on-demand systems are currently being promoted as a viable and cost-effective alternative to existing mobility-on-demand solutions [232]. Today, there are a number of trials around the world for delivering autonomous on-demand mobility. In the majority of cases,

a human backup driver is still required in the vehicle in case of emergencies. In their ultimate form, however, these solutions would not include humans or backup drivers and hence would result in a sharp reduction in travel costs for customers and would allow people to have more flexible mode choices [233–235].

Several recent studies, which relied on millennial surveys, report that younger people are less keen to own private cars. In a study by car sharing company Zipcar, it is reported that half of the millennials interviewed say they would prefer public transport and car sharing systems to privately owned cars [236]. With this in mind, shareable autonomous electric vehicles (particularly those in which electricity is produced through clean resources, e.g., wind turbines or solar systems) appear like a promising proposition for decreasing the overall number of private cars. This would in turn directly address the problems of oil dependency, pollution, promote higher utilisation rates and reduce parking lot sprawls [237]. To date, few studies have dealt with the implications of AMoD systems. Some of the studies of particular relevance to this research are described below.

5.3.1. Lisbon

The Lisbon study [238] examined the potential impacts that would result from the implementation of a shared and fully autonomous vehicle fleet. To perform this assessment, the researchers developed an agent-based model to simulate the behaviour of all entities in the system. Travellers, as potential users of the shared mobility system; cars, which are dynamically routed on the road network to pick-up and drop-off clients, or to move to, from, and between stations; and dispatcher system tasked with efficiently assigning cars to clients while respecting the defined service quality standards, e.g., with regard to waiting time and detour time. The analysis was based on a real urban context, the city of Lisbon, Portugal. The simulation used a representation of the street network, using origin and destination data derived from a fine-grained database of trips based on a detailed travel survey. Trips were allocated to different modes: walking, shared self-driving vehicles, or high-capacity public transport. A set of constraints were established (e.g., that all trips should take at most five minutes longer than today's car trips take for all scenarios, and assumed all trips are done by shared vehicles and none by buses or private cars). The study also modelled a scenario, which included high-capacity public transport (Metro in the case of Lisbon). The study modelled two different car-sharing concepts, "TaxiBots", a term the researchers coined for self-driving vehicles shared simultaneously by several passengers (i.e., ride sharing), and "AutoVots", cars which pick-up and drop-off single passengers sequentially (car sharing). For the different scenarios, the researchers measured the number of cars, kilometres travelled, impacts on congestion, and impacts on parking space. The results indicated that shared self-driving fleets could deliver the same mobility as today with significantly fewer cars. When serviced by ride-sharing TaxiBots and a good underground system, 90% of cars could be removed from the city. Even in the scenario that least reduces the number of cars (AutoVots without underground), nearly half of all cars could be removed without impacting the level of service. Even at peak hours, only about one-third (35%) of today's cars would be needed on the roads (TaxiBots with underground), without reducing overall mobility. On-street parking could be totally removed with a fleet of shared self-driving cars, allowing in a medium-sized European city such as Lisbon, reallocating 1.5 million square metres to other public uses. This equates to almost 20% of the surface of the kerb-to-kerb street area (or 210 football pitches!). These findings suggest that shared self-driving fleets could significantly reduce congestion. In terms of environmental impact, only 2% more vehicles would be needed for a fleet of cleaner, electric, shared self-driving vehicles, to compensate for reduced range and battery charging time.

5.3.2. Stockholm

In the Stockholm study [239], the assessments included both a fleet consisting of currently in use gasoline and diesel cars as well as electric cars. The results showed that an autonomous vehicle-based personal transport system has the potential to provide an on-demand door-to-door transport with a

high level of service, using less than 10% of today's private cars and parking places. In order to provide an environmental benefit and lower congestion, the autonomous vehicle would require users to accept ride-sharing, allowing a maximum 30% increase of their travel time (15% on average) and a start time window of 10 min. In a scenario where users were not inclined to accept a lower level of service, i.e., no ride-sharing and no delay, empty vehicle drive will lead to increased road traffic increasing environmental impacts and congestion. In a scenario which looked at electric cars, an autonomous vehicle-based system and electric vehicle technology seemed to provide a 'perfect' match that could contribute to a sustainable transport system in Stockholm.

5.3.3. Austin

The Austin case study [240] investigated the potential travel and environmental implications of autonomous shared mobility systems by simulating a 12-mile by 24-mile area in Austin, Texas. The multi-agent transport simulation (Matsim) software was used for conducting this experiment using 100,000 randomly drawn person-trips out of 4.5 million Austin's regional trips. The study claimed that each autonomous shared car would almost replace around nine conventional vehicles within the 24-mile by 12-mile area while providing the same level of service, but would generate approximately 8% more vehicle-miles travelled. Their study also confirmed that this system would decrease the emissions by not only replacing the heavier vehicles with higher emissions rates but also by cutting down on the number of cold starts.

5.3.4. New York

The New York case study [241] introduced the expand and target algorithm, which was integrated with three different scheduling strategies for dispatching autonomous vehicles. The study also implemented an agent-based simulation platform and empirically evaluated the proposed approaches using New York City taxi data. Experimental results demonstrated that the algorithms significantly improve passengers' experience by reducing the average passenger waiting time by around 30% and increasing the trip success rate by around 8%.

5.3.5. Melbourne

In a similar study conducted in Australia [242,243], the authors explored the performance of Autonomous Mobility-on-Demand (AMoD) systems under uncertain travel demands in an urban environment using a case study of Melbourne. The results of the simulation model developed for the study showed that an AMoD system could reduce the current fleet size by 84% while still meeting the same demand for travel. This, however, comes at a cost of more vehicle-kilometres Travelled (VKT). The increase in VKT is significant and amounts to around 77% for scenarios in which the vehicles are used in car-sharing systems, and 29% for the scenarios in which vehicles are used as ride-sharing systems. These findings showed that the benefits reported in other studies have mainly been overestimated. In this study, the authors also discovered a strong quadratic relationship between AMoD fleet size and VKT [242].

6. Challenges and Opportunities

It is expected that operating public transport services in a flexible, on-demand manner would lead to higher user acceptance and optimum network utilisation. In order to achieve the above, there are a few critical obstacles to overcome. The main challenge of FMoD service optimisation is to find the trade-off between obtaining the profitability level for operators while maintaining high-quality service for users [19]. To accomplish this, future prediction of passenger demand to the FMoD services, adequate vehicle fleet allocation, and optimum route choice need to be conducted in real-time. Real-time big data analysis for traffic flow prediction is a key challenge [244], which is further complicated by unpredictable human behaviour making a future prediction of users' travel patterns even more challenging.

Other challenges include safety concerns, liability in case of an accident and bootstrapping problems, which is if there are many users for a particular ride-sharing service, there are many opportunities to find users that can share a ride [95]. Ridesharing is generally characterised by the following features: dynamic, independent, cost-sharing, non-recurring and prearranged trips. The main aims of implementing a system like a ride-sharing are the minimisation of system-wide vehicle miles and travel time, and maximisation of the number of participants [245]. Therefore, addressing these issues and improving the system to achieve the objectives are challenging. Furthermore, [245] has discussed three ride-sharing variants, namely single rider-single driver arrangement, single driver-multiple rider arrangement, and single rider-multiple driver arrangement. To obtain optimal arrangements for the above, deep learning theories and advanced mathematical models are needed but these are increasingly being developed with a higher level of detail and sophistication and once well-developed will overcome many of the existing challenges today.

Furthermore, the ride-sharing or ride-hailing services such as Uber and Lyft have also been criticised as a threat to public transport (particularly suburban buses) as these services are cannibalising other forms of conventional public transport modes. The bus industry, in particular, is increasingly being asked to innovate and improve its offerings to be able to compete with other more convenient modes.

The need for further research and in particular to develop a better understanding of the potential impacts of these services on congestion, the environment and the wider social impacts is needed. Further research on user acceptance is also required, particularly circumstances under which users would use such services, especially autonomous solutions in the future, and how they will impact land use changes.

Autonomous solutions introduce some other technical and operational challenges [16,246]:

- Minimum fleet sizing: This relates to determining the minimum number of vehicles required to keep outstanding demands uniformly bounded. Parameters such as arrival rate, average O-D distance, mobility demand distribution, average velocity and average service times are important to consider.
- 2. Performance-driven fleet sizing: The number of vehicles to be used to ensure the quality of service provided to the user is no less than a given threshold.
- 3. The ability of autonomous vehicles to gain environmental awareness to enable reliable, smooth, and safe driving.
- 4. Challenges when connecting intelligent vehicles and infrastructure to provide shared, on-demand service to the customers.
- 5. Software challenges, especially system security and integrity.
- 6. Other social, user acceptance, planning, standards, legislative and insurance challenges with regard to the operation of AMoD

A thorough understanding of the above challenges of FMoD is important to embrace the future of mobility to welcome every potential opportunity.

Addressing the Challenges of Flexible Mobility on Demand (FMoD)

There is strong research momentum today to address the key challenges facing these novel mobility solutions [247]. In particular, there is an increasing focus on the role of vehicle automation in developing next-generation on-demand services. For example, a recent study on using electric vehicles [248] has proposed a fleet optimisation and operational model which was evaluated in simulation.

In another study, an innovative interface has been developed [249] for on-demand bus systems, where users can register or nominate bus stops themselves. In addition, [250,251] described an innovative on-demand bus system development and validity for different city types in Japan. Using machine learning techniques, Q-learning [252] has found an effective pick-up point selection process.

In the operation of the on-demand bus system, buses move only when a passenger requests the service through the internet or mobile phone. The request includes details of pick-up and drop-off location and the desired time of arriving at a destination. This is made possible through the selection of most appropriate nearby vehicles and routing algorithms. The algorithms would first determine which buses should accept the new service request. Then, the routing algorithm is used to identify the route and to schedule based on the new service requests [253].

Furthermore, operational level cost-effective services comprise determination of adequate fleet size followed by optimal operation of buses. Therefore, passenger demand prediction is important to determine bus headways and the required fleet size. Therefore, passenger demand prediction models developed by [254,255] serve as good examples but these need to be supplemented by a deeper understanding of the travel patterns [256].

The authors of this paper are currently engaged in research to develop and evaluate practical models to deliver convenient, efficient, reliable, and door-to-door on-demand transport service. The work is being supplemented by comprehensive simulation models that will be used as test-beds for evaluating the performance of these enhanced algorithms.

7. Policy Insights

The MaaS and shared mobility trends are breaking down the boundaries between different transport modes. This is largely due to the fact that technology is creating an intermediate level between the different means of transport and their users, which is made possible by the fusion of data into a new data layer. This is introducing new challenges to the regulatory and policy frameworks of transport. For the users, the focus will no longer be on the transport mode, but rather on mobility which will increasingly become an information service with physical transportation products, rather than a transportation product with services [257]. In addition to ensuring public safety and access to services, regulators and policymakers must also oversee other aspects such as interconnection, interoperability, capacity management, standards, and security. To achieve this, regulators must focus on the establishment of new comprehensive regulatory frameworks, which would enable the use of information technology, especially across the different transport modes. In particular, this will imply the need to develop an outcome-focused regulatory framework, which (1) puts the users at the centre of the new mobility system; and (2) encourages innovation and promotes safety, security, social equity and environmental sustainability. In the short-term, MaaS will also provide public agencies with the opportunity to bring innovation to their own transport services. In the long term, public agencies may need to rethink their role and consider opportunities for public-private-partnerships and service agreements with private mobility providers (Center for Automated Research, 2016). This new perspective will allow policy maker to focus on the regulation of the new data layers as well as the interface between the data layers and the physical transportation services.

In many cities in the U.S., public agencies are increasingly considering MaaS as an opportunity to provide more mobility options and, also address the first and last kilometre problem, particularly during late hours of the night and in low-density areas. Few cities are already collaborating with mobility providers such as Bridj and Uber to supplement and strengthen their existing public transport offerings and are viewing the technology and mobility providers as partners rather than competitors. Some examples from Tampa, Dallas, Atlanta, Memphis, and San Francisco demonstrate how this can be achievable and cost-effective [71].

Another challenge to policymakers is the protection of privacy. Those who have access to the data (and especially to the way the data is conveyed to the end-users) control the information and have immense power. The abuse of such data and information can result in market distortions, security risks, and diminished privacy protection [257].

The anticipated arrival of shared autonomous mobility-on-demand systems is also likely to introduce major challenges to policymakers. A two-day international workshop was conducted at the

Automated Vehicle Symposium in San Francisco, California in 2016. The aim was to develop a shared automated mobility service by utilising different studies and discussions worldwide.

Table 5 represents a summary of the implications of this technology regarding policies and future research needs [258]. Regulations will in the future play a key role in the emergence and development of autonomous vehicles and the new business models around them such autonomous MaaS. They are also likely to be the biggest hurdle for the deployment and public acceptance of these services. The regulators are already adapting and rethinking their approaches on this issue to avoid stifling the innovative uses of these technologies. Effective responses require an early and on-going dialogue between regulators, developers and the public in which regulators would create legal frameworks that are flexible but robust. An important role for the regulators will be to limit physical risks especially those that might be posed during interim years when legacy fleets of cars would interact with the autonomous vehicles that are offering the MaaS mobility solutions.

Implications in Terms of	Goals	Policy Changes/Solutions	Future Research Needs
Effective automated shared mobility	 Reduce costs, fewer delays, and improves users' experience 	 Automated vehicle lane considerations Funding to adopt these new technologies Implement a single form of payment 	 Payments on all types of shared mobility Implications related to labour and equity Design vehicles to reach the optimal level of services
Safety measures	 Reduce vehicle accidents/collision with other vehicles/ injuries and fatal. Improve safety in terms of preventing crimes Improve safety experience in the vehicles (e.g., harassment, antisocial behaviour, child safety) 	 Implement safety targets and measures. Criteria for designing a vehicle (e.g., clear visibility, emergency button, surveillance) 	 How to ensure safety on pickups points and drop-offs points Technology implementation that aid in avoiding collisions What are the acceptable collison rates? Different cultures present different safety standards
Equity	 Affordable services to all income levels Accessibility near schools, jobs 'Special needs' requirements Provide a free flow of data 	 Require fare integration with equitable fare structures Equal allocations to transport networks to avoid congestion Effective road pricing Multiple pilot studies and testing Ensure a friendly infrastructure for automated vehicles 	 Automated vehicles will lead labour issues? The transition between recent non-automated vehicles and automated vehicles. How to optimise the service/ search for optimisation methods.

Table 5. A summary of the implications of the technology regarding policies and future research needs of shared automated mobility services [258].

8. Summary, Findings, and Future Research Directions

Cities around the world are anticipated to benefit from the use of smart mobility technologies, but they must first overcome a number of challenges. The deployment of these technologies, complemented by appropriate governance and regulatory changes, will deliver substantial benefits [30]. These include user benefits such as personalised smart mobility services tailored to the user's diverse needs and easy access to mobility.

For the public sector, full deployment of technology-related infrastructure would improve the effectiveness of the whole transport system resulting in a more efficient allocation of resources,

improved traffic incident management and a more reliable transport system through advanced data processing.

For businesses, the benefits would include new profitable markets and business models for new transport services; renewed opportunities for the traditional transport and infrastructure business sectors as parts of innovative service concepts and co-operation; and smarter transport connections for all sectors (examples are Uber, Lyft, etc.).

Finally, it is worth noting that the sweeping changes anticipated by disruptive mobility have started to inspire visions of a very different future, as well as a good deal of hype. To distinguish between the hype and reality, it is important going forward to keep abreast of progress and advancements in this space, and to develop informed insights on how these disruptions are likely to change the urban mobility landscape and promote sustainable smart cities.

Findings

This paper provided a comprehensive review of the literature related to flexible mobility on demand systems, and included an environmental scan of the social, environmental, technological and economic impacts. The paper focused on the challenges and opportunities, and the impacts these systems are having on our cities, and success factors for their deployment. The paper first provided a historical overview of the developments in this field, describing applications that ranged from conventional dial-a-ride service to latest applications that rely on digital platforms and mobile apps. The paper also synthesised results from a large body of literature showing the advantages and other impacts of these mobility solutions, and the factors that led to either their failure or success. The paper described the numerous attempts and trials that have focused on changing travel behaviour and shifting travellers from car-dependent environments and encouraging the use of sustainable modes of transport. These studies documented some evidence of instances where users have sold a car or delayed purchasing a new one as a result of wide-spread shared mobility. Some of these findings are summarised in Figure 2 below.



Figure 2. Environmental scan of FMoD systems. Source: Authors.

Despite these findings, more work is still needed, and this paper has identified the research challenges required to enhance the performance of these solutions from the commercial, operational, and public acceptance perspectives. In particular, the paper identified key factors to the success of these systems including the ability of underlying algorithms to predict travel demand patterns under uncertain demand profiles, which are made more complex given unpredictable human behaviour. There is also scope to improve the algorithms used for determination of optimum routes, and fleet and schedule management. To this end, machine learning, Big Data analytics, and large scale optimisation [259] are expected to dominate the field. Finally, and given these challenges, it is proposed that future research is informed by rigorous testing of these systems in simulation environments to test feasibility before commercial deployment in the field.

Author Contributions: The authors' contributions are as follows: Conceptualisation, S.L., H.D., and S.A.B.; Methodology, S.L., H.D., and S.A.B.; Writing—original draft preparation, S.L. and R.A.; Writing—review and editing, H.D. and S.A.B.; Supervision, H.D. and S.A.B.

Funding: This research received no external funding.

Acknowledgments: Sohani Liyanage acknowledges her PhD scholarship provided by the Swinburne University of Technology. Rusul Abduljabbar acknowledges the Iraqi Government for her PhD scholarship.

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations:

Abbreviation	Description
IoT	Internet of Things
FMoD	Flexible mobility on demand
DRT	Demand-responsive transport
DART	Dial a Ride Transit
FTS	Flexible transport services
MaaS	Mobility as a service
MoD	Mobility on demand
AMoD	Autonomous mobility on demand
AI	Artificial intelligence
MAV	Minimum attribute vehicle
STAR	Short term auto rental
B2C	Business to customer
P2P	Peer-to-peer
P2C	Peer-to-customer

References

- 1. Nagurney, A.; Wakolbinger, T. Supernetworks: An introduction to the concept and its applications with a specific focus on knowledge supernetworks. *Int. J. Knowl. Cult. Chang. Manag.* 2005, *4*, 1523–1530. [CrossRef]
- 2. Dia, H. Mapping the value of collaborative mobility. In Proceedings of the 23rd ITS World Congress, Melbourne, Australia, 10–14 October 2016.
- 3. Kaas, H.-W.; Gao, P.; Mohr, D.; Wee, D. *Automotive Revolution—Perspective towards* 2030; McKinsey & Company: New York, NY, USA, 2016.
- 4. Dia, H. Low Carbon Mobility for Future Cities: Principles and Applications; IET: Stevenage, UK, 2017.
- Liddle, B. Urban density and climate change: A STIRPAT analysis using city-level data. *J. Transp. Geogr.* 2013, 28, 22–29. [CrossRef]
- Li, H.; Li, B.; Lu, H. Carbon Dioxide Emissions, Economic Growth, and Selected Types of Fossil Energy Consumption in China: Empirical Evidence from 1965 to 2015. *Sustainability* 2017, 9, 697.
- Liu, Y.; Huang, L.; Kaloudis, A.; Støre-Valen, M. Does urbanization lead to less energy use on road transport? Evidence from municipalities in Norway. *Transp. Res. Part D Transp. Environ.* 2017, *57*, 363–377. [CrossRef]
- Shahbaz, M.; Loganathan, N.; Muzaffar, A.T.; Ahmed, K.; Jabran, M.A. How urbanization affects CO₂ emissions in Malaysia? the application of STIRPAT model. *Renew. Sustain. Energy Rev.* 2016, 57, 83–93. [CrossRef]

- He, Z.; Xu, S.; Shen, W.; Long, R.; Chen, H. Impact of urbanization on energy related CO₂ emission at different development levels: Regional difference in China based on panel estimation. *J. Clean. Prod.* 2017, 140, 1719–1730. [CrossRef]
- 10. Ceder, A.A.; Bagloee, S.A. Transit-network design methodology for actual-size road networks. Methodological. *Transp. Res. Part B Methodol.* **2011**, *45*, 1787–1804.
- 11. Wickham, J.; Lohan, M. *The Transport Rich and the Transport Poor: Car Dependency and Social Class in Four European Cities*; National University of Irelan: Maynooth, Ireland, 1999.
- 12. Mackett, R.L. Increasing car dependency of children: Should we be worried? *Proc. Inst. Civ. Eng. Munic. Eng.* **2002**, 151, 29–38. [CrossRef]
- 13. Buys, L.; Snow, S.; van Megen, K.; Miller, E. Transportation behaviours of older adults: An investigation into car dependency in urban Australia. *Australas. J. Ageing Keywords* **2012**. [CrossRef] [PubMed]
- 14. Pavone, M.; Smith, S.L.; Frazzoli, E.; Rus, D. Robotic Load Balancing for Mobility-on-Demand Systems. *Robot. Sci. Syst.* **2012**, *31*, 839–854. [CrossRef]
- 15. Chong, Z.J.; Qin, B.; Bandyopadhyay, T.; Wongpiromsarn, T.; Rebsamen, B.; Dai, P.; Rankin, E.S.; Ang, M.H. Autonomy for mobility on demand. *Adv. Intell. Syst. Comput.* **2013**, *193*, 671–682.
- Spieser, K.; Treleaven, K.; Zhang, R.; Frazzoli, E.; Morton, D.; Pavone, M. Toward a Systematic Approach to the Design and Evaluation of Automated Mobility-on-Demand Systems: A Case Study in Singapore. In *Road Vehicle Automation*; Springer: Cham, Switzerland, 2014; pp. 1–16.
- 17. Mulley, C.; Nelson, J.D. Flexible transport services: A new market opportunity for public transport. *Res. Transp. Econ.* **2009**, *25*, 39–45. [CrossRef]
- 18. Nelson, J.D.; Wright, S.; Masson, B.; Ambrosino, G.; Naniopoulos, A. Recent developments in Flexible Transport Services. *Res. Transp. Econ.* **2010**, *29*, 243–248. [CrossRef]
- 19. Atasoy, B.; Ikeda, T.; Song, X.; Ben-Akiva, M.E. The concept and impact analysis of a flexible mobility on demand system. *Transp. Res. Part C Emerg. Technol.* **2015**, *56*, 373–392. [CrossRef]
- 20. Khattak, A.J.; Yim, Y. Traveler Response to Innovative Personalized Demand-Responsive Transit in the San Francisco Bay Area. *J. Urban Plan. Dev.* **2004**, *130*, 42–55. [CrossRef]
- 21. Goodwill, J.A.; Carapella, H. *Creative Ways to Manage Paratransit Costs: Final Report, July 2008 [Summary] (No. BD 549 RPWO 28);* National Center for Transit Research (US): Tampa, FL, USA, 2008.
- 22. He, W.; Yan, G.; da Xu, L. Developing vehicular data cloud services in the IoT environment. *IEEE Trans. Ind. Inform.* **2014**, *10*, 1587–1595. [CrossRef]
- 23. Sheehan, R. Mobility on Demand. 2014. Available online: http://innovativemobility.org/wp-content/uploads/2015/02/Robert-Sheehan-Mobility-on-Demand.pdf (accessed on 16 June 2018).
- 24. Mitchell, W.J.; Borroni-Bird, C.; Burns, L.D. *Reinventing the Automobile: Personal Urban Mobility for the 21st Century*; MIT Press: Cambridge, MA, USA, 2010; p. 240.
- 25. Brake, J.; Mulley, C.; Nelson, J.D.; Wright, S. Key lessons learned from recent experience with Flexible Transport Services. *Transp. Policy* **2007**, *14*, 458–466. [CrossRef]
- 26. Choo, C.W. The art of scanning the environment. Bull. Am. Soc. Inf. Sci. Technol. 1999, 25, 21–24. [CrossRef]
- 27. Doherty, M.J.; Sparrow, F.T.; Sinha, K.C. Public use of autos: Mobility enterprise project. *J. Transp. Eng.* **1987**, 113, 84–94. [CrossRef]
- 28. Sulopuisto, O. Why Helsinki's Innovative On-Demand Bus Service Failed. *Citiscope*. 2016. Available online: http://citiscope.org/story/2016/why-helsinkis-innovative-demand-bus-service-failed (accessed on 10 July 2016).
- 29. Bouton, S.; Knupfer, S.; Mihov, I.; Swartz, S. Urban Mobility at a Tipping Point | McKinsey & Company. 2015. Available online: http://www.mckinsey.com/business-functions/sustainability-and-resource-productivity/ our-insights/urban-mobility-at-a-tipping-point (accessed on 10 January 2016).
- Burrows, A.; Bradburn, J. Journeys of the Future. 2015. Available online: http://www.atkinsglobal. com/~{}/media/Files/A/Atkins-Corporate/uk-and-europe/uk-thought-leadership/reports/ Journeysofthefuture_300315.pdf (accessed on 2 August 2016).
- 31. Wollschlaeger, D. *Digital Disruption and the Future of the Automotive Industry;* IBM Center for Applied Insights: Armonk, NY, USA, 2015; p. 16.
- 32. Ford. Ford Smart Mobility Map. *Ford Smart Mobility*. 2015. Available online: https://media.ford. com/content/dam/fordmedia/NorthAmerica/US/2015/03/02/MobilityExperiments.pdf (accessed on 2 August 2016).

- 33. Hietanen, S. Mobility as a Service—European Model of Digital Era Transport. *ITS Finland*. Available online: http://merjakyllonen.fi/merja/wp-content/uploads/2015/10/Hietanen-ITS-Finland.pdf (accessed on 26 August 2018).
- 34. Titcomb, J. UberPool has been Used more than 1 Million Times in London. *The Telegraph*, 7 June 2016.
- 35. Barber, L. Uber Reveals London Ridesharing Figures for UberPool. 2016. Available online: http://www.cityam.com/242760/uber-reveals-london-ride-sharing-figures-for-uberpool (accessed on 6 July 2016).
- 36. Greenfield, A. Helsinki's Ambitious Plan to Make Car Ownership Pointless in 10 Years. *The Guardian*. 2014. Available online: https://www.theguardian.com/cities/2014/jul/10/helsinki-shared-public-transport-plan-car-ownership-pointless (accessed on 2 August 2016).
- 37. Rissanen, K. Helsinki Regional Transport Authority (HSL). 2016. Available online: https://www.hsl.fi/en/ helsinki-regional-transport-authority. (accessed on 6 September 2018).
- 38. Kaufman, R. Chasing the Next Uber—Next City: Which City Will Be the First to Crack on-Demand Mobility? 2016. Available online: https://nextcity.org/features/view/helsinki-kutsuplus-on-demand-transportation-mobility-next-uber (accessed on 17 October 2018).
- 39. Shaheen, S.; Cohen, A.; Zohdy, I. *Shared Mobility: Current Practices and Guiding Principles*; U.S. Department of Transportation Federal Highway Administration: Washington, DC, USA, 2016; p. 120.
- 40. Shaheen, S.; Cohen, A.; Zohdy, I.; Kock, B. *Smartphone Applications to Influence Travel Choices: Practices and Policies*; (No. FHWA-HOP-16-023); U.S. Department of Transportation Federal Highway Administration: Washington, DC, USA, 2016.
- 41. Machado, C.; Hue, N.d.; Berssaneti, F.; Quintanilha, J. An Overview of Shared Mobility. *Sustainability* **2018**, 10, 4342. [CrossRef]
- 42. Bellos, I.; Ferguson, M.; Toktay, L.B. The Car Sharing Economy: Interaction of Business Model Choice and Product Line Design. *Manuf. Serv. Oper. Manag.* **2017**, *19*, 185–201. [CrossRef]
- 43. Dowling, R.; Kent, J. Practice and public-private partnerships in sustainable transport governance: The case of car sharing in Sydney, Australia. *Transp. Policy* **2015**, *40*, 58–64. [CrossRef]
- 44. Baptista, P.; Melo, S.; Rolim, C. Energy, Environmental and Mobility Impacts of Car-sharing Systems. Empirical Results from Lisbon, Portugal. *Procedia Soc. Behav. Sci.* **2014**, *111*, 28–37. [CrossRef]
- 45. Rothenberg, S. Sustainability through Servicizing. 2013. Available online: https://www.researchgate.net/publication/228360037_Sustainability_through_servicizing (accessed on 14 May 2018).
- 46. Cernov, M. The Effect of Environmental Awareness on Car Sales. 2010. Available online: https://www.google.com.hk/url?sa=t&rct=j&q=&esrc=s&source=web&cd=1&ved= 2ahUKEwiEuvfl6tXgAhXAxosBHaQaCfUQFjAAegQICRAC&url=http%3A%2F%2Fwww.etd.ceu.hu% 2F2010%2Fcernov_marina.pdf&usg=AOvVaw32dAtkgJEfme808GMWt7oP (accessed on 23 October 2018).
- 47. Ryden, C.; Morin, E. *Mobility Services for Urban Sustainability: Environmental Assessment*; Moses Rep. WP6; Trivector Traffic AB: Stockholm, Sweden, 2005.
- 48. Cervero, R.; Tsai, Y. City CarShare in San Francisco, California: Second-Year Travel Demand and Car Ownership Impacts. *Transp. Res. Rec. J. Transp. Res. Board* **2004**, *1887*, 117–127. [CrossRef]
- 49. Shaheen, S.; Cohen, A. Growth in Worldwide Carsharing: An International Comparison. *Transp. Res. Rec. J. Transp. Res. Board* **2007**, 1992, 81–89. [CrossRef]
- 50. Katzev, R. Car Sharing: A New Approach to Urban Transportation Problems. *Anal. Soc. Issues Public Policy* **2003**, *3*, 65–86. [CrossRef]
- 51. Tuan Seik, F. Vehicle ownership restraints and car sharing in Singapore. *Habitat Int.* **2000**, *24*, 75–90. [CrossRef]
- 52. Parker, J.; Walker, C.; Johnson, R. What can we learn from car sharing experiences in the UK? *Proc. Inst. Civ. Eng. Transp.* **2011**, *164*, 181–188. [CrossRef]
- 53. Shaheen, S.; Guzman, S.; Zhang, H. Bikesharing in Europe, the Americas, and Asia. *Transp. Res. Rec. J. Transp. Res. Board* **2010**, 159–167. [CrossRef]
- 54. DeMaio, P. Bike-sharing: History, Impacts, Models of Provision, and Future. *J. Public Transp.* **2009**, *12*, 41–56. [CrossRef]
- 55. Midgley, P. The Role of Smart Bike-sharing Systems in Urban Mobility. *Journeys* 2009, *2*, 23–31.
- 56. Meddin, R. The Bike-Sharing World: First Days of Summer 2011. *The Bike-Sharing Blog*, 20 June 2011; viewed 1 May 2015.

- Yang, T.; Haixiao, P.; Qing, S. Bike-sharing systems in Beijing, Shanghai and Hangzhou and their impact on travel behaviour. In Proceedings of the Transportation Research Board Annual Meeting, Washington, DC, USA, 23–27 January 2011.
- 58. Share, A.B. Melbourne Bike Share Survey; Prep. VicRoads: Melbourne, Australia, 2011.
- 59. Shaheen, S.; Zhang, H.; Martin, E.; Guzman, S. China's Hangzhou public bicycle. *Transp. Res. Rec. J. Transp. Res. Board* **2011**, 2247, 33–41. [CrossRef]
- 60. Fishman, E.; Washington, S.; Haworth, N. Bike Share: A Synthesis of the Literature. *Transp. Rev.* **2013**, *33*, 148–165. [CrossRef]
- 61. Consulting, L. Capital Bikeshare 2011; Memb. Surv. Report. 2012. Available online: https://d21xlh2maitm24. cloudfront.net/wdc/Capital-Bikeshare-SurveyReport-Final.pdf?mtime=20161206135935 (accessed on 12 September 2018).
- 62. Transport London. Travel in London; Transport London: London, UK, 2010.
- 63. Saberi, M.; Ghamami, M.; Gu, Y.; Shojaei, M.H.S.; Fishman, E. Understanding the impacts of a public transit disruption on bicycle sharing mobility patterns: A case of Tube strike in London. *J. Transp. Geogr.* **2018**, *66*, 154–166. [CrossRef]
- 64. Fishman, E.; Schepers, P. Global bike share: What the data tells us about road safety. *J. Saf. Res.* **2016**, *56*, 41–45. [CrossRef] [PubMed]
- 65. Fishman, E.; Washington, S.; Haworth, N.; Watson, A. Factors influencing bike share membership: An analysis of Melbourne and Brisbane. *Transp. Res. Part A Policy Pract.* **2014**, *71*, 17–30. [CrossRef]
- 66. Fishman, E.; Schepers, P. The Safety of Bike Share Systems Discussion Paper 168. *Int. Transp. Forum.* Available online: https://www.itf-oecd.org/sites/default/files/docs/safety-bike-share-systems.pdf (accessed on 14 December 2018).
- 67. Schepers, P.; Wolt, K.K.; Fishman, E. The Safety of E-Bikes The Safety of E-Bikes Discussion Paper. *Int. Transp. Forum.* 2018. Available online: https://www.itf-oecd.org/sites/default/files/docs/safety-e-bikes-the-netherlands_0.pdf (accessed on 19 November 2018).
- Demaio, P.; Gifford, J. Will Smart Bikes Succeed as Public Transportation in the United States? *J. Public Transp.* 2004, 7, 1–15. [CrossRef]
- 69. Curran, A. *TransLink Public Bike System Feasibility Study*; PBS Feasibility Study March 2008; Quay Communications Inc.: Vancouver, BC, Canada, 2008.
- Owyang, J. Ten Ways Mobility-as-a-Service Changes Your Lifestyle. Web Strategy LLC. 2015. Available online: http://www.web-strategist.com/blog/2015/09/11/ten-ways-mobility-as-a-service-changes-yourlifestyle/ (accessed on 12 January 2016).
- 71. Center for Automated Research. The Impact of New Mobility Services on the Automotive Industry. 2016. Available online: http://www.cargroup.org/?module=Publications&event=View&pubID=138 (accessed on 2 August 2016).
- Jittrapirom, P.; Caiati, V.; Feneri, A.-M.; Ebrahimigharehbaghi, S.; González, M.J.A.; Narayan, J. Mobility as a Service: A Critical Review of Definitions, Assessments of Schemes, and Key Challenges. *Urban Plan.* 2017, 2, 13. [CrossRef]
- 73. Intelligent Transport Systems. Mobility as a Service: Does Australia Want It? 2018. Available online: https://imovecrc.com/news-articles/personal-public-mobility/maas-australian-readiness/ (accessed on 17 December 2018).
- 74. Krista Huhtala-Jenks, M.F. 'Mobility as a Service'—The new transport paradigm. *Its Transp. Manag. Suppl.* **2014**, *12*, 1–3.
- 75. The Future of Urban Mobility, Digital-Age Transportation. Available online: https://www2.deloitte.com/ insights/us/en/industry/automotive/digital-age-transportation.html (accessed on 23 June 2018).
- 76. Hensher, D.A. Future bus transport contracts under a mobility as a service (MaaS) regime in the digital age: Are they likely to change? *Transp. Res. Part A Policy Pract.* **2017**, *98*, 86–96. [CrossRef]
- 77. Kamargianni, M.; Li, W.; Matyas, M.; Schäfer, A. A Critical Review of New Mobility Services for Urban Transport. *Transp. Res. Procedia* **2016**, *14*, 3294–3303. [CrossRef]
- 78. City Department of Transportation in the City of Brussels, MoBIB. The Different Supports for Our Transport Tickets. Available online: https://www.stib-mivb.be/article.html?l=en&_guid=30af0085-2483-3410-5394a71daf08acd1 (accessed on 24 November 2018).

- 79. GVH (Greater Hannover Transport Association). Hannovermobil. Available online: https://www.gvh.de/ en/mobilitaetsshop/produktuebersicht/hannovermobil/ (accessed on 15 August 2018).
- 80. Urban Innovation Vienna Smart City Wien. *The Mobility Platform of the Future*. 2018. Available online: https://smartcity.wien.gv.at/site/en/smile-2/ (accessed on 27 November 2018).
- 81. Luk, J.; Olszewski, P. Integrated public transport in Singapore and Hong Kong. *Road Transp. Res.* **2003**, *12*, 41–51.
- 82. Finger, M. Mobility-as-a-Services: From the Helsinki Experience to a European Model? 2015. Available online: http://fsr.eui.eu/wp-content/uploads/150309-1-Finger-1.pdf (accessed on 27 July 2018).
- Gould, E.; Wehrmeyer, W.; Leach, M. Transition pathways of e-mobility services. *Wit Trans. Ecol. Environ.* 2015, 194, 349–359.
- Giesecke, R.; Surakka, T.; Hakonen, M. Conceptualising mobility as a service. In Proceedings of the 2016 Eleventh International Conference on Ecological Vehicles and Renewable Energies (EVER), Monte Carlo, Monaco, 6–8 April 2016; pp. 1–11.
- 85. Holmberg, P.-E.; Collado, M.; Sarasini, S.; Williander, M. *Mobility as a Service: Describing the Framework*; RISE Viktoria: Göteborg, Sweden, 2016; pp. 1–54.
- Nemtanu, F.C.; Costea, I.M.; Obreja, L.G. Model of intelligent traffic sensors—Application in hardware in the loop. In Proceedings of the 2017 40th International Spring Seminar on Electronics Technology (ISSE), Sofia, Bulgaria, 10–14 May 2017; pp. 1–5.
- 87. Davidson, P.; Spinoulas, A. Driving alone versus riding together-How shared autonomous vehicles can change the way we drive. *Road Transp. Res. J. Aust. N. Z. Res. Pract.* **2016**, *25*, 51.
- Wong, Y.Z.; Hensher, D.A.; Mulley, C. Emerging Transport Technologies and the Modal Efficiency Framework: A Case for Mobility as a Service (MaaS). 2017. Available online: https://ses.library.usyd.edu.au/bitstream/ 2123/17503/1/Thredbo_15_Thredbo_15_Paper_44.pdf (accessed on 27 August 2018).
- Ebrahimi, S.; Sharmeen, F.; Meurs, H. Innovative Business Architectures (BAs) for Mobility as a Service (MaaS)—Exploration, assessment, and categorization using operational MaaS cases. In Proceedings of the Transportation Research Board 97th Annual Meeting, Washington, DC, USA, 7–11 January 2018.
- 90. Bagloee, S.; Asadi, M.; Richardson, L. Methodology for parking modeling and pricing in traffic impact studies. *Transp. Res. Res. J. Transp. Res. Board* **2012**, *2319*, 1–12. [CrossRef]
- Stenholm, P.; Tapio, P.; McKinnon, A.; Liimatainen, H.; Kallionpää, E.; Pöllänen, M. Decarbonizing road freight in the future—Detailed scenarios of the carbon emissions of Finnish road freight transport in 2030 using a Delphi method approach. *Technol. Forecast. Soc. Chang.* 2014, *81*, 177–191.
- Currie, G. Institute of Transport Studies, ATSE Parliamentary Briefing. 2016. Available online: http://publictransportresearchgroup.info/wp-content/uploads/2017/02/PRESENTATION-2016-ATSE-Parliamentary-Briefing-Melbourne-Transport-Problems-Progress.pdf (accessed on 18 July 2018).
- 93. Roşca, E.; Roşca, M.A.; Valentin, F. Traffic Congestion and Energy Consumption. 2012, pp. 709–714. Available online: http://www.agir.ro/buletine/1458.pdf (accessed on 12 July 2018).
- 94. Taylor, M.A.P.; Smith, N. Modelling the Impacts of Transport Policies to Reduce Greenhouse Gas Emissions from Urban Freight Transport in Sydney. *J. East. Asia Soc. Transp. Stud.* **2005**, *6*, 3135–3150.
- 95. Cici, B.; Markopoulou, A.; Frias-Martinez, E.; Laoutaris, N. Assessing the Potential of Ride-Sharing Using Mobile and Social Data: A Tale of Four Cities. In Proceedings of the ACM International Conference Ubiquitous Computing, Seattle, WA, USA, 13–17 September 2014; pp. 201–211.
- 96. Firnkorn, J.; Müller, M. What will be the environmental effects of new free-floating car-sharing systems? The case of car2go in Ulm. *Ecol. Econ.* **2011**, *70*, 1519–1528. [CrossRef]
- 97. Caulfield, B. Estimating the environmental benefits of ride-sharing: A case study of Dublin. *Transp. Res. Part D Transp. Environ.* **2009**, *14*, 527–531. [CrossRef]
- Shaheen, S.; Cohen, A. Innovative Mobility Carsharing Outlook Carsharing Market Overview, Analysis, and Trends. 2016. Available online: http://innovativemobility.org/wp-content/uploads/2016/02/Innovative-Mobility-Industry-Outlook_World-2016-Final.pdf (accessed on 27 May 2018).
- 99. Burns, L.D.; Scarborough, B.A. Business Plan for a Sustainable Mobility Initiative. 2012. Available online: http://wordpress.ei.columbia.edu/mobility/files/2012/12/Business-Plan-for-a-Sustainable-Mobility-Initiative.pdf (accessed on 17 August 2018).
- Fagnant, D.J.; Kockelman, K.M. The travel and environmental implications of shared autonomous vehicles, using agent-based model scenarios. *Transp. Res. Part C Emerg. Technol.* 2014, 40, 1–13. [CrossRef]

- 101. Hu, X.; Liao, Z.; Wang, J.; He, J. Shared parking policy analysis based on game theory. In Proceedings of the 15th COTA International Conference of Transportation Professionals, Beijing, China, 24–27 July 2015; American Society of Civil Engineers (ASCE): Beijing, China, 2015; pp. 3841–3853.
- 102. World Resources Institute, World Resources 1994–95. 1994. Available online: http://www.wri.org/publication/world-resources-1994-95 (accessed on 24 August 2018).
- 103. Liu, J.; Lin, D.; Li, C. Designing and implementing a service oriented traveler information system. In Proceedings of the Conference on Traffic and Transportation Studies, ICTTS, Kunming, China, 3 August 2010; Volume 383, pp. 741–749.
- 104. Papadimitratos, P.; la Fortelle, A.D.; Paristech, M.; Evenssen, K.; Asa, Q. Vehicular Communication Systems: Enabling Technologies, Applications, and Future Outlook on Intelligent Transportation. *Ieee Commun. Mag.* 2009, 47, 84–95. [CrossRef]
- 105. Manyika, J.; Chui, M.; Bughin, J.; Dobbs, R.; Bisson, P.; Marrs, A. Disruptive Technologies: Advances That Will Transform Life, Business, and the Global Economy; McKinsey Glob. Insitute: San Francisco, CA, USA, 2013; p. 163.
- 106. Salim Bitam, A.M. Cloud Computing for Intelligent Transportation System. *Int. J. Soft Comput. Eng.* **2012**, *2*, 568–572.
- 107. Jaworski, P.; Edwards, T.; Moore, J.; Burnham, K. Cloud computing concept for Intelligent Transportation Systems. In Proceedings of the 2011 14th International IEEE Conference on Intelligent Transportation Systems, Washington, DC, USA, 5–7 October 2011; pp. 391–936.
- 108. Hu, P.; Dhelim, S.; Ning, H.; Qiu, T. Survey on fog computing: Architecture, key technologies, applications and open issues. *J. Netw. Comput. Appl.* **2017**, *98*, 27–42. [CrossRef]
- Butler, B. What Is Fog Computing? Connecting the Cloud to Things. 2018. Available online: https://www.networkworld.com/article/3243111/internet-of-things/what-is-fog-computingconnecting-the-cloud-to-things.html (accessed on 2 January 2019).
- 110. Liu, J.; Li, J.; Zhang, L.; Dai, F.; Zhang, Y.; Meng, X.; Shen, J. Secure intelligent traffic light control using fog computing. *Future Gener. Comput. Syst.* **2018**, *78*, 817–824. [CrossRef]
- 111. Stojmenovic, I.; Wen, S.; Huang, X.; Luan, H. An overview of fog computing and its security issues. *Concurr. Comput. Pract. Exp.* **2016**, *28*, 2991–3005. [CrossRef]
- 112. Hietanen, S. Mobility as a Service—The New Transport Model? *Eurotransport Magasine*. 2014. Available online: http://www.itsineurope.com/its10/media/press_clippings/ITSSupp_et214.pdf (accessed on 3 August 2016).
- 113. Maas Global. Better Than Your Own Car. *MaaS Global*. 2016. Available online: http://maas.global/maas-asa-concept/ (accessed on 4 July 2016).
- 114. Maas Australia, MaaS Australia—Mobility as a Service Australia. *Maas Aus.* 2016. Available online: http://maasaustralia.com/ (accessed on 4 July 2016).
- 115. Australian Transport Assessment and Planning Guidelines. In *T1 Travel Demand Modelling*; 2016. Available online: https://atap.gov.au/tools-techniques/travel-demand-modelling/files/T1_Travel_ Demand_Modelling.pdf (accessed on 30 July 2018).
- 116. Seber, G.A.; Lee, A.J. Linear Regression Analysis; John Wiley Sons: Hoboken, NJ, USA, 2012; Volume 329.
- 117. Jung, J.; Koo, Y. Analyzing the effects of car sharing services on the reduction of greenhouse gas (GHG) emissions. *Sustainability* **2018**, *10*, 539. [CrossRef]
- De Lorimier, A.; El-Geneidy, A.M. Understanding the factors affecting vehicle usage and availability in carsharing networks: A case study of Communauto carsharing system from Montréal, Canada. *Int. J. Sustain. Transp.* 2013, 7, 35–51. [CrossRef]
- 119. Hawas, Y. Simulation-Based Regression Models to Estimate Bus Routes and Network Travel Times. *J. Public Transp.* **2013**, *16*, 107–130. [CrossRef]
- 120. Řehák, B.; Frumar, M.; Koudelka, L. Effect of atmosphere on the electrical conductivity of RuO₂ and ruthenate layers. *J. Mater. Sci. Lett.* **1984**, *3*, 1011–1014. [CrossRef]
- 121. Tran, T.D.; Ovtracht, N.; D'Arcier, B.F. Modeling bike sharing system using built environment factors. *Procedia Cirp* **2015**, *30*, 293–298. [CrossRef]
- 122. Varagouli, E.G.; Simos, T.E.; Xeidakis, G.S. Fitting a multiple regression line to travel demand forecasting: The case of the prefecture of Xanthi, Northern Greece. *Math. Comput. Model.* **2005**, *42*, 817–836. [CrossRef]

- 123. Cao, Y.; Wang, J. The Key Contributing Factors of Customized Shuttle Bus in Rush Hour: A Case Study in Harbin City. *Procedia Eng.* **2016**, *137*, 478–486. [CrossRef]
- 124. Davison, L.; Enoch, M.; Ryley, T.; Quddus, M.; Wang, C. A survey of demand responsive transport in great Britain. *Transp. Policy* **2014**, *31*, 47–54. [CrossRef]
- 125. Mitchell, T.M. Artificial Neural Networks. Machine Learning. 1997. Available online: http://profsite.um.ac. ir/~{}monsefi/machine-learning/pdf/Machine-Learning-Tom-Mitchell.pdf (accessed on 19 August 2018).
- 126. Little, T.D.; Wei, W.W.S. Time Series Analysis; Oxford University Press: Oxford, UK, 2013.
- 127. Menard, S. (Ed.) *Handbook of Longitudinal Research: Design, Measurement, and Analysis;* Elsevier: Amsterdam, The Netherlands, 2007; Volume 136.
- 128. Kaltenbrunner, A.; Meza, R.; Grivolla, J.; Codina, J.; Banchs, R. Urban cycles and mobility patterns: Exploring and predicting trends in a bicycle-based public transport system. *Pervasive Mob. Comput.* 2010, *6*, 455–466. [CrossRef]
- 129. Gaudry, M. An aggregate time-series analysis of urban transit demand: The Montreal case. *Transp. Res.* **1975**, *9*, 249–258. [CrossRef]
- 130. Lane, B.W. A time-series analysis of gasoline prices and public transportation in US metropolitan areas. *J. Transp. Geogr.* **2012**, *22*, 221–235. [CrossRef]
- 131. Andrade, T.C.; de Arruda Pereira, M.; Wanner, E.F. Development of an Application Using a Clustering Algorithm for Definition of Collective Transportation Routes and Times. *GeoInfo.* 2014, pp. 13–24. Available online: http://www.geoinfo.info/proceedings_geoinfo2014.split/Paper03-F-p19.pdf (accessed on 14 May 2018).
- Galba, T.; Balkić, Z.; Martinović, G. Public transportation bigdata clustering. *Int. J. Electr. Comput. Eng. Syst.* 2013, 4, 21–26.
- 133. Briand, A.S.; Côme, E.; Trépanier, M.; Oukhellou, L. Analyzing year-to-year changes in public transport passenger behaviour using smart card data. *Transp. Res. Part C Emerg. Technol.* 2017, 79, 274–289. [CrossRef]
- 134. Sun, F.; Dubey, A.; White, J.; Gokhale, A. Transit-hub: A smart public transportation decision support system with multi-timescale analytical services. *Clust. Comput.* **2017**, 1–16. [CrossRef]
- Agard, B.; Morency, C.; Trépanier, M. Mining public transport user behaviour from smart card data. In Proceedings of the 12th IFAC Symp, Saint Malo, France, 17–19 May 2006; pp. 1–14.
- 136. Lathia, N.; Smith, C.; Froehlich, J.; Capra, L. Individuals among commuters: Building personalised transport information services from fare collection systems. *Pervasive Mob. Comput.* **2013**, *9*, 643–664. [CrossRef]
- 137. Nin, J.; Carrera, D.; Villatoro, D. On the Use of Social Trajectory-Based Clustering Methods for Public Transport Optimization; Citizen in Sensor Networks; Lect. Notes Comput. Sci.; Springer: Cham, Switzerland, 2014; Volume 8313.
- 138. Trépanier, M.; Habib, K.M.; Morency, C. Are transit users loyal? Revelations from a hazard model based on smart card data. *Can. J. Civ. Eng.* **2012**, *39*, 610–618. [CrossRef]
- 139. Ester, M.; Kriegel, H.P.; Sander, J.; Xu, X. A density-based algorithm for discovering clusters in large spatial databases with noise. In Proceedings of the KDD '96 Proceedings of the Second International Conference on Knowledge Discovery and Data Mining, Portland, OR, USA, 2–4 August 1996; pp. 226–231.
- 140. Kieu, L.M.; Bhaskar, A.; Chung, E. Mining temporal and spatial travel regularities for transit planning. In Proceedings of the Australasian Transport Research Forum 2013, Queensland University of Technology, Brisbane, Australia, 2–4 October 2013.
- 141. Morency, C.; Trépanier, M.; Agard, B. Analysing the variability of transit users behaviour with smart card data. In Proceedings of the IEEE ITSC, Toronto, ON, Canada, 17–20 September 2006; pp. 44–49.
- 142. Bhaskar, A.; Chung, E. Passenger segmentation using smart card data. *IEEE Trans. Intell. Transp. Syst.* 2015, 16, 1537–1548.
- 143. Tan, P.N.; Steinbach, M.; Kumar, V. Introduction to Data Mining; Pearson Educ.: London, UK, 2006; p. 769.
- 144. Lindner, A.; Pitombo, C.S.; Cunha, A.L. Estimating motorized travel mode choice using classifiers: An application for high-dimensional multicollinear data. *Travel Behav. Soc.* **2017**, *6*, 100–109. [CrossRef]
- 145. Rashidi, T.H.; Abbasi, A.; Maghrebi, M.; Hasan, S.; Waller, T.S. Exploring the capacity of social media data for modelling travel behaviour: Opportunities and challenges. *Transp. Res. Part C Emerg. Technol.* 2017, 75, 197–211. [CrossRef]
- 146. Abduljabbar, R.; Dia, H.; Liyanage, S.; Bagloee, S. Applications of Artificial Intelligence in Transport: An Overview. *Sustainability* **2019**, *11*, 189. [CrossRef]

- 147. Cagliero, L.; Cerquitelli, T.; Chiusano, S.; Garza, P.; Xiao, X. Predicting critical conditions in bicycle sharing systems. *Computing* **2017**, *99*, 39–57. [CrossRef]
- 148. El Mahrsi, M.K.; Come, E.; Oukhellou, L.; Verleysen, M. Clustering Smart Card Data for Urban Mobility Analysis. *Ieee Trans. Intell. Transp. Syst.* **2017**, *18*, 712–728. [CrossRef]
- Ceapa, I.; Smith, C.; Capra, L. Avoiding the crowds: Understanding tube station congestion patterns from trip data. In Proceedings of the ACM SIGKDD Int. Work. Urban Comput., Beijing, China, 12 August 2012; pp. 134–141.
- 150. Klema, V.; Laub, A. The singular value decomposition: Its computation and some applications. *IEEE Trans. Autom. Control* **1980**, 25, 164–176. [CrossRef]
- 151. Hoyer, P.O. Non-negative matrix factorization with sparseness constraints. J. Mach. Learn. Res. 2004, 5, 1457–1469.
- 152. Zhao, J.; Xie, X.; Xu, X.; Sun, S. Multi-view learning overview: Recent progress and new challenges. *Inf. Fusion* **2017**, *38*, 43–54. [CrossRef]
- 153. PourMohammadBagher, L.; Ebadzadeh, M.M.; Safabakhsh, R. Graphical model based continuous estimation of distribution algorithm. *Appl. Soft Comput.* **2017**, *58*, 388–400. [CrossRef]
- 154. Guinness, R.E. Beyond where to how: A machine learning approach for sensing mobility contexts using smartphone sensors. *Sensors* **2015**, *15*, 9962–9985. [CrossRef] [PubMed]
- 155. Suthaharan, S. Machine learning models and algorithms for big data classification. In *Integrated Series in Information Systems*; Springer: Boston, MA, USA, 2016.
- 156. Liaw, A.; Wiener, M. Classification and Regression by randomForest. R News 2002, 2, 18–22.
- 157. Goodall, W.; Fishman, T.D.; Bornstein, J.; Bonthron, B. The rise of mobility as a service: Reshaping how urbanites get around. *Delotitte Rev.* **2017**, *20*, 111–130.
- 158. Della Corte, V.; D'Andrea, C.; Savastano, I.; Zamparelli, P. Smart Cities and Destination Management: Impacts and Opportunities for Tourism Competitiveness. *Eur. J. Tour. Res.* **2017**, *17*, 7–27.
- Schulz, T.; Rockmann, R.; Weeger, A. Service Composition in Networks–Towards a Typology of Intermediaries. 2016. Available online: https://aisel.aisnet.org/amcis2016/DigitalComm/Presentations/ 26/ (accessed on 1 July 2018).
- 160. Karlsson, I.C.M.; Sochor, J.; Strömberg, H. Developing the 'Service' in Mobility as a Service: Experiences from a Field Trial of an Innovative Travel Brokerage. *Transp. Res. Procedia* **2016**, *14*, 3265–3273. [CrossRef]
- 161. IDC Corporate USA. Big Data & Analytics; IDC: Framingham, MA, USA, 2016; p. 1.
- 162. Russom, P. *High-Performance Data Warehousing*; TDWI Best Practices Report; TDWI Res.: Renton, WA, USA, 2012.
- 163. Russom, P.; Lavalle, S.; Lesser, E.; Shockley, R.; Hopkins, M.S.; Kruschwitz, N. Big Data Analytics. In Proceedings of the ASIST 2011, New Orleans, LA, USA, 9–12 October 2011; Volume 19, p. 40.
- 164. Zakir, J.; Seymour, T.; Berg, K. Big Data Analytics. *Issues Inf. Syst.* 2015, Volume 16, pp. 81–90. Available online: http://www.iacis.org/iis/2015/2_iis_2015_81-90.pdf (accessed on 10 July 2018).
- 165. Khan, Z.; Anjum, A.; Kiani, S.L. Cloud based big data analytics for smart future cities. In Proceedings of the 2013 IEEE/ACM 6th International. Conference on Utility and Cloud Computing, UCC 2013, Dresden, Germany, 9–12 December 2013; pp. 381–386.
- 166. Hashem, I.A.T.; Chang, V.; Anuar, N.B.; Adewole, K.; Yaqoob, I.; Gani, A.; Ahmed, E.; Chiroma, H. The role of big data in smart city. *Int. J. Inf. Manag.* **2016**, *36*, 748–758. [CrossRef]
- Weise, T.; Chiong, R.; Tang, K. Evolutionary optimization: Pitfalls and booby traps. J. Comput. Sci. Technol. 2012, 27, 907–936. [CrossRef]
- Komodakis, N.; Pesquet, J.C. Playing with duality: An overview of recent primal? dual approaches for solving large-scale optimization problems. *IEEE Signal Process. Mag.* 2015, 32, 31–54. [CrossRef]
- Omidvar, M.N.; Member, S.; Li, X.; Member, S. Cooperative Co-Evolution With Differential Grouping for Large Scale Optimization. *IEEE Trans. Evol. Comput.* 2014, 18, 378–393. [CrossRef]
- 170. Bell, J.E.; McMullen, P.R. Ant colony optimization techniques for the vehicle routing problem. *Adv. Eng. Inform.* 2004, *18*, 41–48. [CrossRef]
- 171. Abdmouleh, Z.; Gastli, A.; Ben-Brahim, L.; Haouari, M.; Al-Emadi, N.A. Review of optimization techniques applied for the integration of distributed generation from renewable energy sources. *Renew. Energy* 2017, 113, 266–280. [CrossRef]

- 172. Bottou, L.; Curtis, F.E.; Nocedal, J. Optimization Methods for Large-Scale Machine Learning. *Siam Rev.* 2018, 60, 223–311. [CrossRef]
- 173. Weber, S.; Nagy, A.; Schüle, T.; Schnörr, C.; Kuba, A. A Benchmark Evaluation of Large-Scale Optimization Approaches to Binary Tomography. In Proceedings of the Discrete Geometry for Computer Imagery, DGCI 2006, Szeged, Hungary, 25–27 October 2006; Lecture Notes in Computer Science 2006. Volume 4245, pp. 146–156.
- 174. Ferreira, S.L.C.; Lemos, V.A.; de Carvalho, V.S.; da Silva, E.G.; Queiroz, A.F.; Felix, C.S.; da Silva, D.L.F.; Dourado, G.B.; Oliveira, R.V. Multivariate optimization techniques in analytical chemistry—An overview. *Microchem. J.* 2018, 140, 176–182. [CrossRef]
- 175. Hannah, L.A. Stochastic Optimization. 2006, pp. 1–20. Available online: http://www.stat.columbia.edu/ ~liam/teaching/compstat-spr14/lauren-notes.pdf (accessed on 23 May 2018).
- Shapiro, A.; Homem-De-Mello, T.; Kim, J. Conditioning of convex piecewise linear stochastic programs. *Math. Program. Ser. B* 2002, 94, 1–19. [CrossRef]
- 177. Robinson, S.M. Analysis of Sample-Path Optimization. Math. Oper. Res. 1996, 21, 513–528. [CrossRef]
- 178. Robbins, H.; Monro, S. A Stochastic Approximation Method. Ann. Math. Stat. 1951, 22, 400–407. [CrossRef]
- 179. Downing, D.J.; Gardner, R.H.; Hoffman, F.O. An examination of response-surface methodologies for uncertainty analysis in assessment models. *Technometrics* **1985**, *27*, 151–163. [CrossRef]
- 180. Lim, E. Response surface computation via simulation in the presence of convexity. In Proceedings of the Winter Simulation Conference, Baltimore, MD, USA, 5–8 December 2010; pp. 1246–1254.
- Kleijnen, J.P.C.; Sargent, R.G. A methodology for fitting and validating metamodels in simulation. *Eur. J.* Oper. Res. 2000, 120, 14–29. [CrossRef]
- 182. Zhu, J.; Ye, X. Development of destination choice model with pairwise district-level constants using taxi GPS data. *Transp. Res. Part C Emerg. Technol.* **2018**, 93, 410–424. [CrossRef]
- 183. Yang, X.-S. Metaheuristic Optimization: Algorithm Analysis and Open. In Proceedings of the 10th International Symposium on Experimental Algorithms (SEA 2011), Crete, Greece, 5–7 May 2011; pp. 21–32.
- 184. Xhafa, F.; Abraham, A. Computational models and heuristic methods for Grid scheduling problems. *Future Gener. Comput. Syst.* 2010, 26, 608–621. [CrossRef]
- Malcolm, S.A.; Zenios, S.A. Robust optimization for power systems capacity expansion under uncertainty. J. Oper. Res. Soc. 1994, 45, 1040–1049. [CrossRef]
- Melvey, J.M.; Vanderbei, R.J.; Zenios, S.A. Robust Optimization of Large-Scale Systems John. *Oper. Res.* 2015, 43, 264–281. [CrossRef]
- Schneider, M.H.; Zenios, S.A. A Comparative Study of Algorithms for Matrix Balancing. *Oper. Res.* 1990, 38, 439–455. [CrossRef]
- 188. Yan, Y.; Meng, Q.; Wang, S.; Guo, X. Robust optimization model of schedule design for a fixed bus route. *Transp. Res. Part C Emerg. Technol.* **2012**, *25*, 113–121. [CrossRef]
- 189. Yan, Y.; Liu, Z.; Meng, Q.; Jiang, Y. Robust Optimization Model of Bus Transit Network Design with Stochastic Travel Time. *J. Transp. Eng.* **2013**, *139*, 625–634. [CrossRef]
- 190. Zhang, X.Y.; Yin, F.; Zhang, Y.M.; Liu, C.L.; Bengio, Y. Drawing and Recognizing Chinese Characters with Recurrent Neural Network. *IEEE Trans. Pattern Anal. Mach. Intell.* **2018**, 40, 849–862. [CrossRef] [PubMed]
- 191. Du, K.L.; Swamy, M.N.S. Neural Networks and Statistical Analyses; Springer: London, UK, 2013.
- 192. Caterini, A.L.; Chang, D.E. *Deep Neural Networks in a Mathematical Framework*; Springer: Cham, Switzerland, 2018.
- Hochreiter, S.; Schmidhuber, J.U. Long Short-Term Memory. Neural Comput. 1997, 9, 1735–1780. [CrossRef] [PubMed]
- 194. Krizhevsky, A.; Sutskever, I.; Hinton, G.E. ImageNet Classification with Deep Convolutional Neural Networks. *Adv. Neural Inf. Process. Syst.* 2012, *6*, 1–9. [CrossRef]
- Kim, P. MATLAB Deep Learning. 2017. Available online: https://link.springer.com/content/pdf/10.1007/ 978-1-4842-2845-6.pdf (accessed on 28 July 2018).
- McCann, M.T.; Jin, K.H.; Unser, M. Deep Convolutional Neural Network for Inverse Problems in Imaging. *Ieee Signal Process. Mag.* 2017, 34, 85–95. [CrossRef]
- 197. Sun, Z.J.; Xue, L.; Xu, Y.M.; Wang, Z. Overview of deep learning. Appl. Res. Comput. 2012, 29, 2806–2810.
- Deng, L.; Yu, D. Deep learning: Methods and applications. *Found. Trends Sig. Process.* 2013, 7, 197–387. [CrossRef]

- 199. LeCun, Y.; Bengio, Y.; Hinton, G. Deep learning. Nature 2015, 521, 436–444. [CrossRef] [PubMed]
- 200. Silver, D.; Huang, A.; Maddison, C.J.; Guez, A.; Sifre, L.; Van Den Driessche, G.; Schrittwieser, J.; Antonoglou, I.; Panneershelvam, V.; Lanctot, M.; et al. Mastering the game of Go with deep neural networks and tree search. *Nature* **2016**, *529*, 484–489. [CrossRef] [PubMed]
- 201. Bagloee, S.A.; Ceder, A.; Bozic, C. Effectiveness of en route traffic information in developing countries using conventional discrete choice and neural-network models. *J. Adv. Transp.* **2014**, *48*, 486–506. [CrossRef]
- 202. Schmidhuber, J. Deep learning in neural networks: An overview. *Neural Netw.* **2015**, *61*, 85–117. [CrossRef] [PubMed]
- 203. Anthes, G. Deep learning comes of age. Commun. ACM 2013, 56, 13–15. [CrossRef]
- 204. Arel, I.; Rose, D.C.; Karnowski, T.P. Deep machine learning-a new frontier in artificial intelligence research. *Ieee Comput. Intell. Mag.* **2010**, *5*, 13–18. [CrossRef]
- 205. Zhou, D.Y.; Zhang, J.Z. Direct ridership forecast model of urban rail transit stations based on spatial weighted LS-SVM. *J. China Railw. Soc.* **2014**, *36*, 1–7.
- 206. Liu, L.; Chen, R.C. A novel passenger flow prediction model using deep learning methods. *Transp. Res. Part C Emerg. Technol.* **2017**, *84*, 74–91. [CrossRef]
- 207. Jung, J.; Sohn, K. Deep-learning architecture to forecast destinations of bus passengers from entry-only smart-card data. *Iet Intell. Transp. Syst.* 2017, *11*, 334–339. [CrossRef]
- Sallab, A.E.; Abdou, M.; Perot, E.; Yogamani, S. Deep reinforcement learning framework for autonomous driving. *Electron. Imaging* 2017, 2017, 70–76. [CrossRef]
- 209. Ben-Elia, E.; Ishaq, R.; Shiftan, Y. If only I had taken the other road: Regret, risk and reinforced learning in informed route-choice. *Transportation* **2013**, *40*, 269–293. [CrossRef]
- 210. Mirchevska, B.; Blum, M.; Louis, L.; Boedecker, J.; Werling, M. Reinforcement learning for autonomous maneuvering in highway scenarios. *Workshop Driv. Assist. Syst. Auton. Driving.* **2017**, *11*, 32–41.
- 211. Thiagarajan, A.; Ravindranath, L.; LaCurts, K.; Madden, S.; Balakrishnan, H.; Toledo, S.; Eriksson, J. VTrack: Accurate, energy-aware road traffic delay estimation using mobile phones. In Proceedings of the 7th ACM Conference on Embedded Networked Sensor Systems, Berkeley, CA, USA, 4–6 November 2009; pp. 85–98.
- 212. Chan, N.D.; Shaheen, S.A. Ridesharing in North America: Past, Present, and Future. *Transp. Rev.* 2012, 32, 93–112. [CrossRef]
- 213. The Commuter Information Source for Maryland, Virginia. How Ridesourcing Services Fit into the Transportation Landscape. 2015, pp. 1–6. Available online: http://www.commuterconnections.org/wp-content/uploads/Newsletter-Winter-2015.pdf (accessed on 25 August 2018).
- 214. Uber. 2018. Available online: https://www.uber.com/au/en/ (accessed on 2 January 2019).
- 215. Lyft. 2018. Available online: https://www.lyft.com/ (accessed on 2 January 2019).
- 216. DiDi. 2018. Available online: https://www.didiglobal.com/ (accessed on 2 January 2019).
- 217. Careem. 2019. Available online: https://www.careem.com/ (accessed on 2 January 2019).
- 218. Lifthero, Lift-Hero. 2019. Available online: https://www.crunchbase.com/organization/lift-hero (accessed on 2 January 2019).
- 219. Hopskipdrive, Hop Skip Drive. 2018. Available online: https://www.hopskipdrive.com/ (accessed on 2 January 2019).
- Correia, G.; Viegas, J.M. Carpooling and carpool clubs: Clarifying concepts and assessing value enhancement possibilities through a Stated Preference web survey in Lisbon, Portugal. *Transp. Res. Part A Policy Pract.* 2011, 45, 81–90. [CrossRef]
- 221. Galland, S.; Knapen, L.; Gaud, N.; Janssens, D.; Lamotte, O.; Koukam, A.; Wets, G. Multi-agent simulation of individual mobility behavior in carpooling. *Transp. Res. Part C Emerg. Technol.* **2014**, 45, 83–98. [CrossRef]
- 222. Teal, R.F. Carpooling: Who, how and why. Transp. Res. Part A Gen. 1987, 21, 203–214. [CrossRef]
- 223. Inturri, G.; Giuffrida, N.; Ignaccolo, M.; le Pira, M.; Pluchino, A.; Rapisarda, A. Testing Demand Responsive Shared Transport Services via Agent-Based Simulations. In *New Trends in Emerging Complex Real Life Problems*; Daniele, P., Scrimali, L., Eds.; AIRO Springer Series: Cham, Switzerland, 2018.
- 224. Stein, D.M. Scheduling Dial-a-Ride Transportation Systems. Available online: https://pubsonline.informs. org/doi/abs/10.1287/trsc.12.3.232 (accessed on 27 October 2018).
- 225. Diana, M. The Importance of Information Flows Temporal Attributes for the Efficient Scheduling of Dynamic Demand Responsive Transport Services. *J. Adv. Transp.* **2006**, *40*, 23–46. [CrossRef]

- Rosin, J. Optibus Uses Artificial Intelligence to Improve Mass Transit's On-Time Performance and Prevent Delays. 2018. Available online: https://finance.yahoo.com/news/optibus-uses-artificial-intelligenceimprove-110000554.html?guccounter=1 (accessed on 27 October 2018).
- 227. Ma, J.; Yang, Y.; Guan, W.; Wang, F.; Liu, T.; Tu, W.; Song, C. Large-scale demand driven design of a customized bus network: A methodological framework and beijing case study. *J. Adv. Transp.* 2017, 2017. [CrossRef]
- 228. Liu, T.; Ceder, A. Analysis of a new public-transport-service concept: Customized bus in China. *Transp. Policy* **2015**, *39*, 63–76. [CrossRef]
- 229. Bridj. 2018. Available online: https://www.bridj.com/journey/ (accessed on 30 December 2018).
- 230. Citylab. Why Helsinki's On-Demand Bus Service Failed. 2016. Available online: https://www.citylab.com/transportation/2016/03/helsinki-on-demand-bus-service-kutsuplus/472545/ (accessed on 27 December 2018).
- 231. NSW Transport. On Demand Public Transport. 2018. Available online: https://transportnsw.info/travelinfo/ways-to-get-around/on-demand-public-transport (accessed on 17 December 2018).
- 232. Pavone, M. Autonomous Mobility-on-Demand Systems for Future Urban Mobility. In *Autonomes Fahren;* Springer: Berlin/Heidelberg, Germany, 2015.
- 233. Shen, Y.; Zhang, H.; Zhao, J. Integrating shared autonomous vehicle in public transportation system: A supply-side simulation of the first-mile service in Singapore. *Transp. Res. Part A Policy Pract.* 2018, 113, 125–136. [CrossRef]
- 234. Van den Berg, V.A.C.; Verhoef, E.T. Autonomous cars and dynamic bottleneck congestion: The effects on capacity, value of time and preference heterogeneity. *Transp. Res. Part B Methodol.* 2016, 94, 43–60. [CrossRef]
- 235. Dial, R.B. Autonomous dial-a-ride transit introductory overview. *Transp. Res. Part C Emer. Technol.* **1995**, *3*, 261–275. [CrossRef]
- 236. Zipcar. Zipcar's Annual Millennial Survey Shows the Kids Are All Right. 2014. Available online: http://www.zipcar.com/press/releases/fourthannual-millennial-survey (accessed on 3 September 2015).
- 237. Zhang, R.; Spieser, K.; Frazzoli, E.; Pavone, M. Models, Algorithms, and Evaluation for Autonomous Mobility-On-Demand Systems. In Proceedings of the IEEE American Control Conference (ACC), Chicago, IL, USA, 1–3 July 2015; pp. 2573–2587.
- 238. International Transport Forum. Urban Mobility System Upgrade: How Shared Self-Driving Cars Could Change City Traffic. *ITF*. 2015. Available online: http://www.internationaltransportforum.org/Pub/pdf/ 15CPB_Self-drivingcars.pdf (accessed on 10 September 2015).
- 239. Rigole, P. Study of a Shared Autonomous Vehicles Based Mobility Solution in Stockholm. *Master of Science Thesis*. 2014. Available online: http://bit.ly/1Qig7Cu (accessed on 25 August 2015).
- 240. Fagnant, D.J.; Kockelman, K.M.; Bansal, P. Operations of shared autonomous vehicle fleet for austin, texas, market. *Transp. Res. Res. J. Transp. Res. Board* 2015, 2536, 98–106. [CrossRef]
- 241. Shen, W.; Lopes, C. Managing Autonomous Mobility on Demand Systems for Better Passenger Experience. Principles and Practice of Multi-Agent Systems. *Springer International Publishing*. 2015. Available online: http://bit.ly/1nC1WyN (accessed on 11 January 2016).
- 242. Javanshour, F.; Dia, H. Development and evaluation of models for autonomous shared mobility-on-demand systems. In Proceedings of the 34th Conference of the Australian Institutes of Transport Research (CAITR), Brisbane, Australia, 11–12 February 2016.
- 243. Javanshour, F.; Dia, H.; Duncan, G. Exploring the performance of autonomous mobility on-demand systems under demand uncertainty. *Transp. A Transp. Sci.* **2018**, *39*, 1–24. [CrossRef]
- 244. Lv, Y.; Duan, Y.; Kang, W.; Li, Z.; Wang, F.Y. Traffic Flow Prediction With Big Data: A Deep Learning Approach. *Ieee Trans. Intell. Transp. Syst.* 2014, *16*, 865–873. [CrossRef]
- 245. Agatz, N.; Erera, A.; Savelsbergh, M.; Wang, X. Optimization for dynamic ride-sharing: A review. *Eur. J. Oper. Res.* **2012**, 223, 295–303. [CrossRef]
- 246. Bagloee, S.A.; Tavana, M.; Asadi, M.; Oliver, T. Autonomous vehicles: Challenges, opportunities, and future implications for transportation policies. *J. Mod. Transp.* **2016**, *24*, 284–303. [CrossRef]
- 247. Frost. Analysis of the Global on Demand Bus Transit Market. 2016. Available online: https://ww2.frost. com/frost-perspectives/analysis-global-demand-bus-transit-market/ (accessed on 14 December 2018).

- Kawamura, K.; Mukai, N. Optimization of Transport Plan for On-Demand Bus System Using Electrical Vehicles. In *Knowledge-Based and Intelligent Information and Engineering Systems*; Springer: Berlin/Heidelberg, Germany, 2009; pp. 656–663.
- 249. Tsubouchi, K.; Yamato, H.; Hiekata, K. Development and evaluation of new interface for registration of new bus stops for the on-demand bus system. *Int. J. Intell. Transp. Syst. Res.* **2010**, *3*, 188–200. [CrossRef]
- 250. Tsubouchi, K.; Yamato, H.; Hiekata, K. Innovative on-demand bus system in Japan. *IET Intell. Transp. Syst.* **2010**, *4*, 270. [CrossRef]
- 251. Raymond, R.; Sugiura, T.; Tsubouchi, K. Location recommendation based on location history and spatio-temporal correlations for an on-demand bus system. In Proceedings of the 19th ACM SIGSPATIAL International Conference on Advances in Geographic Information Systems, Chicago, IL, USA, 1–4 November 2011; p. 377.
- Mukai, N.; Watanabe, T.; Feng, J. Route Optimization Using Q-Learning for On-Demand Bus Systems. In Knowledge-Based and Intelligent Information and Engineering Systems; Springer: Berlin/Heidelberg, Germany, 2008; pp. 567–574.
- 253. Tsubouchi, K.; Yamato, H. Adapted Scheduling Algorithm for On-Demand Bus System. *Int. J. Intell. Transp. Syst. Res.* **2012**, *10*, 82–92. [CrossRef]
- 254. Zhou, C.; Dai, P.; Li, R. The passenger demand prediction model on bus networks. In Proceedings of the. IEEE 13th International Conference on Data Mining Workshops, ICDMW 2013, Dallas, TX, USA, 7–10 December 2013; pp. 1069–1076.
- 255. Zhou, C.; Dai, P.; Zhang, Z. Passenger demand prediction on bus services. In Proceedings of the 2015 International Conference on Green Computing and Internet of Things, ICGCIoT 2015, Noida, India, 8–10 October 2015; pp. 1430–1435.
- 256. Enoch, M.P.; Parkhurst, G.P. Why do demand responsive transport systems fail? In Proceedings of the Transportation Research Board 85th Annual Meeting, Washington DC, USA, 22–26 January 2006.
- 257. Finger, E.M.; Bert, N.; Kupfer, D. Mobility-as-a-Service: From the Helsinki Experiment to a European Model? FSR Transport. 2015. Available online: http://fsr.eui.eu/Documents/WorkshopPaper/Transport/2015/ 150309MaaSObserver.pdf (accessed on 27 November 2018).
- 258. Lazarus, J.; Shaheen, S.; Young, S.E.; Fagnant, D.; Voege, T.; Baumgardner, W.; Fishelson, J.; Lott, J.S. Shared Automated Mobility and Public Transport. In *Road Vehicle Automation*; Springer: Cham, Switzerland, 2018; Volume 4, pp. 141–161.
- 259. Bagloee, S.A.; Asadi, M.; Sarvi, M.; Patriksson, M. A hybrid machine-learning and optimization method to solve bi-level problems. *Expert Syst. Appl.* **2018**, *95*, 142–152. [CrossRef]



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).